\providecommand{\bysame}{\leavevmode\hbox to3em{\hrulefill}\thinspace}
\providecommand{\MR}{\relax\ifhmode\unskip\space\fi MR }
% \MRhref is called by the amsart/book/proc definition of \MR.
\providecommand{\MRhref}[2]{%
  \href{http://www.ams.org/mathscinet-getitem?mr=#1}{#2}
}
\providecommand{\href}[2]{#2}
\begin{thebibliography}{BGSW22}

\bibitem[AGJP22]{AGJP22}
Massoud Amini, Nasser Golestani, Saeid Jamali, and N.~Christopher Phillips,
  \emph{Group actions on simple tracially $\mathcal{Z}$-absorbing
  {C}*-algebras}, arXiv preprint arXiv:2204.03615 (2022).

\bibitem[BGSW22]{BGSW22}
Joan Bosa, James Gabe, Aidan Sims, and Stuart White, \emph{The nuclear
  dimension of $\mathcal{O}_\infty$-stable {C}*-algebras}, Adv. Math.
  \textbf{401} (2022), Paper No. 108250, 51. \MR{4392219}

\bibitem[Bis90]{Bisch90}
Dietmar~H. Bisch, \emph{On the existence of central sequences in subfactors},
  Trans. Amer. Math. Soc. \textbf{321} (1990), no.~1, 117--128. \MR{1005075}

\bibitem[Bis94]{Bisch94}
\bysame, \emph{Central sequences in subfactors. {II}}, Proc. Amer. Math. Soc.
  \textbf{121} (1994), no.~3, 725--731. \MR{1209417}

\bibitem[BO08]{BrownOzawa}
Nathanial~P. Brown and Narutaka Ozawa, \emph{{C}*-algebras and
  finite-dimensional approximations}, Graduate Studies in Mathematics, vol.~88,
  American Mathematical Society, Providence, RI, 2008. \MR{2391387}

\bibitem[BSKR93]{BSKR93}
Ola Bratteli, Erling St{\o}rmer, Akitaka Kishimoto, and Mikael R{\o}rdam,
  \emph{The crossed product of a {UHF} algebra by a shift}, Ergodic Theory
  Dynam. Systems \textbf{13} (1993), no.~4, 615--626. \MR{1257025}

\bibitem[Bur10]{Burstein10}
Richard~D. Burstein, \emph{Commuting square subfactors and central sequences},
  Internat. J. Math. \textbf{21} (2010), no.~1, 117--131. \MR{2642989}

\bibitem[Con75]{Connes75}
Alain Connes, \emph{Classification of automorphisms of hyperfinite factors of
  type {II}$_1$ and {II}$_{\infty}$ and application to type {III} factors},
  Bull. Amer. Math. Soc. \textbf{81} (1975), no.~6, 1090--1092. \MR{388117}

\bibitem[Con76]{Connes76}
\bysame, \emph{Classification of injective factors cases {II}$_1$,
  {II}$_\infty$, {III}$_\lambda$, $\lambda \neq 1$}, Ann. of Math. (2)
  \textbf{104} (1976), no.~1, 73--115. \MR{454659}

\bibitem[CS19]{CameronSmith19}
Jan Cameron and Roger~R. Smith, \emph{A {G}alois correspondence for reduced
  crossed products of simple {C}*-algebras by discrete groups}, Canad. J. Math.
  \textbf{71} (2019), no.~5, 1103--1125. \MR{4010423}

\bibitem[Cun77]{Cuntz77}
Joachim Cuntz, \emph{Simple {C}*-algebras generated by isometries}, Comm. Math.
  Phys. \textbf{57} (1977), no.~2, 173--185. \MR{467330}

\bibitem[Dav96]{DavidsonBook1}
Kenneth~R. Davidson, \emph{{C}*-algebras by example}, Fields Institute
  Monographs, vol.~6, American Mathematical Society, Providence, RI, 1996.
  \MR{1402012}

\bibitem[DW09]{DadarlatWinter09}
Marius Dadarlat and Wilhelm Winter, \emph{On the {KK}-theory of strongly
  self-absorbing {C}*-algebras}, Math. Scand. \textbf{104} (2009), no.~1,
  95--107. \MR{2498373}

\bibitem[ER21]{EchterhoffRordam21}
Siegfried Echterhoff and Mikael R{\o}rdam, \emph{Inclusions of {C}*-algebras
  arising from fixed-point algebras}, arXiv preprint arXiv:2108.08832 (2021).

\bibitem[EST23]{EndersSchemaitatTikuisis23}
Dominic Enders, Andr{\'e} Schemaitat, and Aaron Tikuisis, \emph{Corrigendum to
  ``{$K$}-theoretic characterization of {C}*-algebras with approximately inner
  flip''}, arXiv preprint arXiv:2303.11106 (2023).

\bibitem[EV00]{ElliottVilladsen00}
George~A. Elliott and Jesper Villadsen, \emph{Perforated ordered
  {$K_0$}-groups}, Canad. J. Math. \textbf{52} (2000), no.~6, 1164--1191.
  \MR{1794301}

\bibitem[Gab19]{Gabe19}
James Gabe, \emph{Classification of $\mathcal{O}_\infty$-stable {C}*-algebras},
  arXiv preprint arXiv:1910.06504 (2019).

\bibitem[Gab20]{Gabe20}
\bysame, \emph{A new proof of {K}irchberg's $\mathcal{O}_2$-stable
  classification}, J. Reine Angew. Math. \textbf{761} (2020), 247--289.
  \MR{4080250}

\bibitem[GH18]{GardellaHirshberg18}
Eusebio Gardella and Ilan Hirshberg, \emph{Strongly outer actions of amenable
  groups on $\mathcal{Z}$-stable {C}*-algebras}, arXiv preprint
  arXiv:1811.00447 (2018).

\bibitem[HO13]{HirshbergOrovitz13}
Ilan Hirshberg and Joav Orovitz, \emph{Tracially $\mathcal{Z}$-absorbing
  {C}*-algebras}, J. Funct. Anal. \textbf{265} (2013), no.~5, 765--785.
  \MR{3063095}

\bibitem[HRW07]{HirshbergRordamWinter07}
Ilan Hirshberg, Mikael R{\o}rdam, and Wilhelm Winter,
  \emph{$\mathcal{C}_0({X})$-algebras, stability and strongly self-absorbing
  {C}*-algebras}, Math. Ann. \textbf{339} (2007), no.~3, 695--732. \MR{2336064}

\bibitem[HW41]{HurewiczWallman}
Witold Hurewicz and Henry Wallman, \emph{Dimension theory}, Princeton
  Mathematical Series, vol. 4, Princeton University Press, Princeton, N. J.,
  1941. \MR{0006493}

\bibitem[HW07]{HirshbergWinter07}
Ilan Hirshberg and Wilhelm Winter, \emph{{R}okhlin actions and self-absorbing
  {C}*-algebras}, Pacific J. Math. \textbf{233} (2007), no.~1, 125--143.
  \MR{2366371}

\bibitem[Izu02]{Izumi02}
Masaki Izumi, \emph{Inclusions of simple {C}*-algebras}, J. Reine Angew. Math.
  \textbf{547} (2002), 97--138.

\bibitem[Izu04]{Izumi04}
\bysame, \emph{Finite group actions on {C}*-algebras with the {R}ohlin
  property. {I}}, Duke Math. J. \textbf{122} (2004), no.~2, 233--280.
  \MR{2053753}

\bibitem[JS99]{JiangSu99}
Xinhui Jiang and Hongbing Su, \emph{On a simple unital projectionless
  {C}*-algebra}, Amer. J. Math. \textbf{121} (1999), no.~2, 359--413.
  \MR{1680321}

\bibitem[Kaw99]{Kawamuro99}
Keiko Kawamuro, \emph{Central sequence subfactors and double commutant
  properties}, Internat. J. Math. \textbf{10} (1999), no.~1, 53--77.
  \MR{1678538}

\bibitem[Kir06]{Kirchberg06}
Eberhard Kirchberg, \emph{Central sequences in {C}*-algebras and strongly
  purely infinite algebras}, Operator {A}lgebras: {T}he {A}bel {S}ymposium
  2004, Abel Symp., vol.~1, Springer, Berlin, 2006, pp.~175--231. \MR{2265050}

\bibitem[KP00]{KirchbergPhillips00}
Eberhard Kirchberg and N.~Christopher Phillips, \emph{Embedding of exact
  {C}*-algebras in the {C}untz algebra $\mathcal{O}_2$}, J. Reine Angew. Math.
  \textbf{525} (2000), 17--53. \MR{1780426}

\bibitem[McD69]{McDuff69}
Dusa McDuff, \emph{Uncountably many {II}$_1$ factors}, Ann. of Math. (2)
  \textbf{90} (1969), 372--377. \MR{259625}

\bibitem[Mor09]{deLacerdaMortari09}
Fernando de~Lacerda Mortari, \emph{Tracial state spaces of higher stable rank
  simple {C}*-algebras}, ProQuest LLC, Ann Arbor, MI, 2009, Thesis
  (Ph.D.)--University of Toronto (Canada). \MR{2753146}

\bibitem[MS12]{MatuiSato12}
Hiroki Matui and Yasuhiko Sato, \emph{Strict comparison and
  $\mathcal{Z}$-absorption of nuclear {C}*-algebras}, Acta Math. \textbf{209}
  (2012), no.~1, 179--196. \MR{2979512}

\bibitem[MvN43]{MvNIV}
Francis~J. Murray and John von Neumann, \emph{On rings of operators. {IV}},
  Ann. of Math. (2) \textbf{44} (1943), 716--808. \MR{9096}

\bibitem[OT18]{OsakaTeruya18}
Hiroyuki Osaka and Tamotsu Teruya, \emph{The {J}iang--{S}u absorption for
  inclusions of unital {C}*-algebras}, Canad. J. Math. \textbf{70} (2018),
  no.~2, 400--425. \MR{3759005}

\bibitem[Pop89]{Popa89}
Sorin Popa, \emph{Sousfacteurs, actions des groupes et cohomologie}, C. R.
  Acad. Sci. Paris S\'{e}r. I Math. \textbf{309} (1989), no.~12, 771--776.
  \MR{1054961}

\bibitem[Pop00]{Popa00}
\bysame, \emph{On the relative {D}ixmier property for inclusions of
  {C}*-algebras}, J. Funct. Anal. \textbf{171} (2000), no.~1, 139--154.
  \MR{1742862}

\bibitem[Rae05]{RaeburnBook}
Iain Raeburn, \emph{Graph algebras}, CBMS Regional Conference Series in
  Mathematics, vol. 103, Published for the Conference Board of the Mathematical
  Sciences, Washington, DC; by the American Mathematical Society, Providence,
  RI, 2005. \MR{2135030}

\bibitem[R{\o}r91]{Rordam91}
Mikael R{\o}rdam, \emph{On the structure of simple {C}*-algebras tensored with
  a {UHF}-algebra}, J. Funct. Anal. \textbf{100} (1991), no.~1, 1--17.
  \MR{1124289}

\bibitem[R{\o}r92]{Rordam92}
\bysame, \emph{On the structure of simple {C}*-algebras tensored with a
  {UHF}-algebra. {II}}, J. Funct. Anal. \textbf{107} (1992), no.~2, 255--269.
  \MR{1172023}

\bibitem[R{\o}r95]{Rordam95}
\bysame, \emph{Classification of certain infinite simple {C}*-algebras}, J.
  Funct. Anal. \textbf{131} (1995), no.~2, 415--458. \MR{1345038}

\bibitem[R{\o}r02]{RordamBook}
\bysame, \emph{Classification of nuclear, simple {C}*-algebras}, Classification
  of nuclear {C}*-algebras. {E}ntropy in operator algebras, Encyclopaedia Math.
  Sci., vol. 126, Springer, Berlin, 2002, pp.~1--145. \MR{1878882}

\bibitem[R{\o}r04]{Rordam04}
\bysame, \emph{The stable and the real rank of $\mathcal{Z}$-absorbing
  {C}*-algebras}, Internat. J. Math. \textbf{15} (2004), no.~10, 1065--1084.
  \MR{2106263}

\bibitem[R{\o}r21]{Rordam21}
\bysame, \emph{Irreducible inclusions of simple {C}*-algebras}, arXiv preprint
  arXiv:2105.11899 (2021).

\bibitem[RS87]{RosenbergSchochet87}
Jonathan Rosenberg and Claude Schochet, \emph{The {K}\"{u}nneth theorem and the
  universal coefficient theorem for {K}asparov's generalized {$K$}-functor},
  Duke Math. J. \textbf{55} (1987), no.~2, 431--474. \MR{894590}

\bibitem[RW10]{RordamWinter10}
Mikael R{\o}rdam and Wilhelm Winter, \emph{The {J}iang-{S}u algebra revisited},
  J. Reine Angew. Math. \textbf{642} (2010), 129--155. \MR{2658184}

\bibitem[Sat10]{Sato10}
Yasuhiko Sato, \emph{The {R}ohlin property for automorphisms of the
  {J}iang-{S}u algebra}, J. Funct. Anal. \textbf{259} (2010), no.~2, 453--476.
  \MR{2644109}

\bibitem[Sch20]{Schafhauser20}
Christopher Schafhauser, \emph{Subalgebras of simple {AF}-algebras}, Ann. of
  Math. (2) \textbf{192} (2020), no.~2, 309--352. \MR{4151079}

\bibitem[Sch22]{Schemaitat22}
Andr\'{e} Schemaitat, \emph{The {J}iang-{S}u algebra is strongly self-absorbing
  revisited}, J. Funct. Anal. \textbf{282} (2022), no.~6, Paper No. 109347, 39.
  \MR{4360358}

\bibitem[Tik12]{Tikuisis12}
Aaron Tikuisis, \emph{Regularity for stably projectionless, simple
  {C}*-algebras}, J. Funct. Anal. \textbf{263} (2012), no.~5, 1382--1407.
  \MR{2943734}

\bibitem[Tik16]{Tikuisis16}
\bysame, \emph{{K}-theoretic characterization of {C}*-algebras with
  approximately inner flip}, Int. Math. Res. Not. IMRN (2016), no.~18,
  5670--5694. \MR{3567256}

\bibitem[Tom05]{Toms05}
Andrew~S. Toms, \emph{On the independence of {$K$}-theory and stable rank for
  simple {C}*-algebras}, J. Reine Angew. Math. \textbf{578} (2005), 185--199.
  \MR{2113894}

\bibitem[Tom08a]{Toms08b}
\bysame, \emph{An infinite family of non-isomorphic {C}*-algebras with
  identical {$K$}-theory}, Trans. Amer. Math. Soc. \textbf{360} (2008), no.~10,
  5343--5354. \MR{2415076}

\bibitem[Tom08b]{Toms08}
\bysame, \emph{On the classification problem for nuclear {C}*-algebras}, Ann.
  of Math. (2) \textbf{167} (2008), no.~3, 1029--1044. \MR{2415391}

\bibitem[TW07]{TomsWinter07}
Andrew~S. Toms and Wilhelm Winter, \emph{Strongly self-absorbing
  {C}*-algebras}, Trans. Amer. Math. Soc. \textbf{359} (2007), no.~8,
  3999--4029. \MR{2302521}

\bibitem[TW08]{TomsWinter08}
\bysame, \emph{{$\mathcal{Z}$}-stable {ASH} algebras}, Canad. J. Math.
  \textbf{60} (2008), no.~3, 703--720. \MR{2414961}

\bibitem[TW09]{TomsWinter09}
\bysame, \emph{The {E}lliott conjecture for {V}illadsen algebras of the first
  type}, J. Funct. Anal. \textbf{256} (2009), no.~5, 1311--1340. \MR{2490221}

\bibitem[TWW15]{TomsWhiteWinter15}
Andrew~S. Toms, Stuart White, and Wilhelm Winter, \emph{$\mathcal{Z}$-stability
  and finite-dimensional tracial boundaries}, Int. Math. Res. Not. IMRN (2015),
  no.~10, 2702--2727. \MR{3352253}

\bibitem[Vil98]{Villadsen98}
Jesper Villadsen, \emph{Simple {C}*-algebras with perforation}, J. Funct. Anal.
  \textbf{154} (1998), no.~1, 110--116. \MR{1616504}

\bibitem[Vil99]{Villadsen99}
\bysame, \emph{On the stable rank of simple {C}*-algebras}, J. Amer. Math. Soc.
  \textbf{12} (1999), no.~4, 1091--1102. \MR{1691013}

\end{thebibliography}

\documentclass[12pt]{amsart}

\usepackage[margin=1.5in]{geometry} 

\include{preabmle}
\include{macros}
\include{letterfonts}

\begin{document}
\title{Tensorially absorbing inclusions of C*-algebras}
\author{Pawel Sarkowicz}
\email{\href{mailto:psark007@uottawa.ca}{psark007@uottawa.ca}}
\address{Department of Mathematics and Statistics, University of Ottawa, 75 Laurier Ave. East, Ottawa, ON, K1N 6N5 Canada}

  \begin{abstract}
    When $\mathcal{D}$ is strongly self-absorbing we say an inclusion $B \subseteq A$ is $\mathcal{D}$-stable if it is isomorphic to the inclusion $B \otimes \mathcal{D} \subseteq A \otimes \mathcal{D}$. We give ultrapower characterizations and show that if a unital inclusion is $\mathcal{D}$-stable, then $\mathcal{D}$-stability can be exhibited for countably many intermediate C*-algebras concurrently. We show that such embeddings between $\mathcal{D}$-stable C*-algebras are point-norm dense in the set of all embeddings, and that every embedding between $\mathcal{D}$-stable C*-algebras is approximately unitarily equivalent to a $\mathcal{D}$-stable embedding. Examples are provided. 
  \end{abstract}

  \maketitle
  \tableofcontents
  
  
\section{Introduction}
  The study of inclusions of C*-algebras has been of recent interest. There is no short supply of research concerning inclusions relating to non-commutative dynamics \cite{Popa00,Izumi02,CameronSmith19,OsakaTeruya18,EchterhoffRordam21}, as well as inclusions of simple C*-algebras \cite{Rordam21}. We discuss inclusions from the lens of tensorially absorbing a strongly self-absorbing C*-algebra $\cD$ \cite{TomsWinter07}.

  When speaking of tensorial absorption with a strongly self-absorbing C*-algebra, central sequences play an imperative role akin to McDuff's character\hyp{}ization of when a $\II_1$ von Neumann algebra absorbs the unique hyperfinite $\II_1$ factor $\cR$ \cite{McDuff69}. Central sequences have been studied since the inception of operator algebras as Murray and von Neumann used them to exhibit non-isomorphic $\II_1$ factors by showing that $\cL(\bF_2)$ does not have property $\Gamma$ \cite{MvNIV}. They were also used in Connes' theorem concerning the uniqueness of $\cR$ \cite{Connes76}, and the classification of automorphisms on hyperfinite factors \cite{Connes75,Connes76}. In \cite{Bisch90,Bisch94}, Bisch considered the central sequence algebra $\cN^\omega \cap \cM'$ associated to an (irreducible) inclusion of $\II_1$ factors $\cN \subseteq \cM$ and characterized when there was an isomorphism $\Phi: \cM \simeq \cM \ov{\otimes} \cR$ such that $\Phi(\cN) = \cN \ov{\otimes} \cR$ in terms of there being non-commuting sequences in $\cN$ which asymptotically commute with the larger von Neumann algebra $\cM$ (in the $\|\cdot\|_2$-norm). As pointed out by Izumi \cite{Izumi04}, there are similar central characterizations for unital inclusions of separable C*-algebras which tensorially absorb a strongly self-absorbing C*-algebra $\cD$ (it was at least pointed out for $\cD$ being one of $M_{n^{\infty}},\cO_2,\cO_{\infty}$).

  For a strongly self-absorbing C*-algebra $\cD$ \cite[Definition 1.3(iv)]{TomsWinter07}, we study $\cD$-stable
  inclusions (see Section \ref{section:dse-embeddings} for detailed definitions), analogous to Bisch's notion for an (irreducible) inclusion of $\II_1$ factors \cite{Bisch90}. We say that an inclusion $B \subseteq A$ is $\cD$-stable if there is an isomorphism $A \simeq A \otimes \cD$ such that
\begin{equation}
  \begin{tikzcd}
A \arrow[r, "\simeq"]                 & A \otimes \cD                 \\
B \arrow[r, "\simeq"] \arrow[u, hook, "\iota"] & B \otimes \cD \arrow[u, hook, "\iota \otimes \id_\cD"']
\end{tikzcd}
\end{equation}
commutes.

We study such inclusions systematically, discussing central sequence char\hyp{}acterizations, permanence properties, and giving examples towards the end. We list some key findings here.  The first is that $\cD$-stable inclusions exist between $\cD$-stable C*-algebras if there is any inclusion, and that the set of $\cD$-stable inclusions is quite large. Moreover, as far as classification of embeddings up to approximate unitary equivalence (in particular by $K$-theory and traces), $\cD$-stable embeddings are all that matter. 

\begin{result}[Proposition \ref{allembeddingsareue}, Corollary \ref{pointnormdensity}]
    Let $A,B$ be unital, separable, $\cD$-stable C*-algebras.
      \begin{enumerate}
        \item The set of $\cD$-stable embeddings $B \into A$ is point-norm dense in the set of all embeddings $B \into A$.
        \item Every embedding $B \into A$ is approximately unitarily equivalent to a $\cD$-stable embedding. 
      \end{enumerate}
  \end{result}

  We note that this set is however not everything. We provide examples of non-$\cD$-stable inclusions of $\cD$-stable C*-algebras, namely by fitting a C*-algebra with perforated Cuntz semigroup or with higher stable rank (in partic\hyp{}ular non-$\cZ$-stable C*-algebras) in between two $\cD$-stable C*-algebras. The second useful tool is that a $\cD$-stable inclusion allows one to find an appropriate isomorphism witnessing $\cD$-stability of countably many intermediate subalge\hyp{}bras at once.


\begin{result}[Theorem \ref{countablymanysubalgebras}]
    Let $B \subseteq A$ be a unital, $\cD$-stable inclusion of separable C*-algebras. If $(C_n)$ is a sequence of C*-algebras such that $B \subseteq C_n \subseteq A$ unitally for all $n$, then there exists an isomorphism $\Phi: A \simeq A \otimes \cD$ such that
      \begin{enumerate}
        \item $\Phi(B) = B \otimes \cD$ and 
        \item $\Phi(C_n) = C_n \otimes \cD$ for all $n \in \bN$. 
      \end{enumerate}
  \end{result}
This is not a trivial condition, as it is not true that any such isomorphism sends every intermediate C*-algebra to its tensor product with $\cD$ (see Example \ref{noteveryintermediate}). In fact, one can always find an intermediate C*-algebra $C$ between $B$ and  $A$ and an isomorphism $A \simeq A \otimes \cD$ sending $B$ to $B \otimes \cD$ which does not send $C$ to $C \otimes \cD$ (although, of course, we will still have $C \simeq C \otimes \cD$). 

The above result, together with the Galois correspondence of Izumi \cite{Izumi02}, allows us to get a result similar to the main theorem of \cite{AGJP22}. There they prove that if $G \act^\alpha A$ is an action of a finite group with the weak tracial Rokhlin property on a C*-algebra $A$ with sufficient regularity conditions, then every C*-algebra between $A^\alpha \subseteq A$ and $A \subseteq A \rtimes_\alpha G$ is $\cZ$-stable. Assuming we have a unital C*-algebra with the same regularity conditions, we show that we can witness $\cZ$-stability of all such intermediate C*-algebras concurrently. 


  \begin{resultcor}[Corollary \ref{weaktracialRokhlinintermediate}]
            Let $A$ be a unital, simple, separable, nuclear $\cZ$-stable C*-algebra and $G \act^\alpha A$ be an action of a finite group with the weak tracial Rokhlin property. There exists an isomorphism $\Phi: A \rtimes_\alpha G \simeq (A \rtimes_\alpha G) \otimes \cZ$ such that whenever $C$ is a unital C*-algebra satisfying either
              \begin{enumerate}
                \item $A^\alpha \subseteq C \subseteq A$ or
                \item $A \subseteq C \subseteq A \rtimes_\alpha G$,
              \end{enumerate}
              we have $\Phi(C) = C \otimes \cZ$. 
          \end{resultcor}

  This paper is structured as follows. We discuss various local properties in Section \ref{section:dse-localembeddings}, and then formalize the notion of a $\cD$-stable embedding in Section \ref{section:dse-embeddings}, examining several properties and consequences. In Section \ref{section:dse-crossed-products} we show how several examples arising from non-commutative dynamical systems fit into the framework of $\cD$-stable inclusions. We finish with several examples in Section \ref{section:dse-examples}.

  %Acknowledgements section
%  \addtocontents{toc}{\protect\setcounter{tocdepth}{0}}
%  \section*{Acknowledgements}
%  Many thanks to my supervisors Thierry Giordano and Aaron Tikuisis for many helpful discussions. 

  %go back to normal TOC ordering..
%  \addtocontents{toc}{\protect\setcounter{tocdepth}{1}}

  %Acknowledgements section
  \addtocontents{toc}{\protect\setcounter{tocdepth}{0}}
  \section*{Acknowledgements}
  Many thanks to my supervisors Thierry Giordano and Aaron Tikuisis for many helpful discussions, as well as to Eusebio Gardella for many helpful comments.  

  %go back to normal TOC ordering..
  \addtocontents{toc}{\protect\setcounter{tocdepth}{1}}
  
  %\renewcommand*{\thetheorem}{\arabic{section}.\arabic{theorem}}

  
    \section{Preliminaries}\label{preliminaries}

  \subsection{Notation} We will use capital letters $A,B,C,D$ to denote C*-algebras and usually a calligraphic $\cD$ to denote a strongly self-absorbing C*-algebra. Generally small letters $a,b,c,d,\dots,x,y,z$ will denote operators in C*-algebras. $A_+$ will denote cone of positive elements in a C*-algebra $A$. If $\ee > 0$ and $a,b$ are elements in a C*-algebra, we will write
\begin{equation}
   a \approx_\ee b
\end{equation}
  to mean that $\|a - b\| < \ee$. This will make some approximations more legible. 

% do we need the algebraic tensor product? So far, no.
  The symbol $\otimes$ will denote the minimal tensor product of C*-algebras, while $\odot$ will mean the algebraic tensor product. We use the minimal tensor product throughout, and it is common for us to deal with nuclear C*-algebras so there should not be any ambiguity. The symbol $\ov{\otimes}$ will denote the von Neumann tensor product. 

  We will denote by $M_n$ the C*-algebra of $n\times n$ matrices, and $M_{n^{\infty}}$ the uniformly hyperfinite (UHF) C*-algebra associated to the supernatural number $n^{\infty}$. We will write $\cQ$ for the universal UHF algebra $\cQ = \bigotimes_{n \in \bN} M_n$. 

  By $G \act^\alpha A$, we will mean that the (discrete) group $G$ acts on $A$ by automorphisms, i.e., $\alpha: G \to \Aut(A)$ is a homomorphism. $A \rtimes_{r,\alpha}G$ will denote the reduced crossed product, which we will just write as $A \rtimes_\alpha G$ if it is clear from context that the group is amenable and $A$ is nuclear (e.g., if $G$ is finite). We will denote by $A^\alpha$ the fixed point subalgebra of the action (or $A^G$ if the action is clear from context).


\subsection{Ultrapowers, central sequences and central sequence algebras}\label{section:ultrapowers}

Fix a free ultrafilter $\omega \in \beta \bN$. Throughout we will use ultrapowers to describe asymptotic behaviour, rather than sequences algebras. This comes down to a matter of taste and one can swap between the two if one so desires, as we will provide local characterizations as well. This also means that all of what we do will be independent of the specific ultrafilter $\omega$.

For a C*-algebra $A$, the ultrapower of $A$ is the C*-algebra
  \begin{equation}
 A_\omega := \ell^{\infty}(A)/c_{0,\omega}(A), 
  \end{equation}
where $c_{0,\omega} := \{(a_n) \in \ell^{\infty}(A) \mid \lim_{n \to \omega} \|a_n\| = 0\}$ is the ideal of $\omega$-null sequences. We can embed $A$ into $A_\omega$ canonically by means of constant sequences: we identify $a \in A$ with the equivalence class of the constant sequence $(a)_n$.

To ease notation, we will usually write elements of $A_\omega$ as sequences $(a_n)$, keeping in mind that these are equivalence classes without explicitly stating it every time. We note that the norm on $A_\omega$ is given by $\|(a_n)\| = \lim_{n \to \omega} \|a_n\|$. 

Kirchberg's $\ee$-test (\cite{Kirchberg06}, Lemma A.1) is essentially the operator algebraists' {\L}o{\' s}' theorem without having to turn to (continuous) model theory. Heuristically, it says that if certain things can be done approximately in an ultrapower, then certain things can be done exactly in an ultrapower.

\begin{lemma}[Kirchberg's $\ee$-test]\label{lemma:Kirchberg-eptest}
Let $(X_n)_n$ be a sequence of sets and suppose that for each $n$, there is a sequence $(f_n^{(k)})$ of functions $f_n^{(k)}: X_n \to [0,\infty)$. For $k \in \bN$, let
\begin{equation}
f_\omega^k(s_1,s_2,\dots) := \lim_{n \to \omega} f_n^{(k)}(s_n). 
\end{equation}
Suppose that for every $m \in \bN$ and $\ee > 0$, there is $s \in \prod_n X_n$ with $f_\omega^{(k)}(s) < \ee$ for $k=1,\dots,m$. Then there exists $t \in \prod_nX_n$ with $f_\omega^{(k)}(t) = 0$ for all $k \in \bN$. 
\begin{comment}
Moreover, there is a sequence $n_1 < n_2 < \cdots$ in $\bN$ such that there are $s_l \in X_{n_l}$ with $f_{n_l}^{(k)} < 2^{-l}$ for $k \leq l, l \in \bN$. \end{comment}
\end{lemma}

The above is useful, although if one so wishes, one can usually construct exact objects from approximate objects by using standard diagonalization arguments (under some separability assumptions).
These sorts of arguments work in both the ultrapower setting and the sequence algebra setting.

Finally, if $\alpha \in \Aut(A)$ is an automorphism, there is an induced automorphism on $A_\omega$, which we will denote by $\alpha_\omega$ given by
\begin{equation}
\alpha_\omega((a_n)) := (\alpha(a_n)). 
\end{equation}

    \subsection{Central sequences and central sequence subalgebras} For a unital C*-algebra $A$, the C*-algebra of $\omega$-central sequences is
\begin{equation}
    A_\omega \cap A' = \{x \in A_\omega \mid [x,a] = 0 \text{ for all } a \in A\},
\end{equation}
    where we are identifying $A \subseteq A_\omega$ with the constant sequences. If $B \subseteq A$ is a unital C*-subalgebra and $S \subseteq A_\omega$ is a subset, we can associate the relative commutant of $S$ in $B_\omega$:
\begin{equation}
    B_\omega \cap S' = \{b \in B_\omega \mid [b,s] = 0 \text{ for all } s \in S\}. 
\end{equation}
    Of particular interest will be when $S = A$, and $B \subseteq A$ is a unital inclusion of separable C*-algebras.

\subsection{Strongly self-absorbing C*-algebras}\label{section:ssa-C*-algebras}

A unital separable C*-algebra $\cD$ is strongly self-absorbing if $\cD \not\simeq \bC$ and there is an isomorphism $\phi: \cD \to \cD \otimes \cD$ which is approximately unitarily equivalent to the first factor embedding $d \mapsto d \otimes 1_\cD$ (see \cite{TomsWinter07}). All known strongly self-absorbing C*-algebras are: the Jiang-Su algebra $\cZ$ \cite{JiangSu99}, the Cuntz algebras $\cO_2$ and $\cO_{\infty}$ \cite{Cuntz77}, UHF algebras of infinite type, and $\cO_{\infty}$ tensor a UHF algebra of infinite type. Strongly self-absorbing C*-algebras have approximately inner flip, and therefore there are $K$-theoretic restrictions on strongly self-absorbing C*-alge\hyp{}bras -- see \cite{Tikuisis16,EndersSchemaitatTikuisis23}.

Tensorial absorption with strongly self-absorbing C*-algebras gives rise to many regular properties, for examples in terms of $K$-theory, traces, and the Cuntz semigroup \cite{JiangSu99,Rordam91,Rordam92,Rordam04}. Of paramount interest is the  Jiang-Su algebra $\cZ$. A cumulation of work has successfully classified all (unital) separable, simple, nuclear, infinite-dimensional, $\cZ$-stable C*-algebras satisfying the Universal Coefficient Theorem (UCT) of Rosenberg and Schochet \cite{RosenbergSchochet87} by means of $K$-theory and traces. We describe how one might work with $\cZ$-stability in terms of its standard building blocks. Recall that, for $n,m \geq 2$, the dimension drop algebras are
  \begin{equation}
\cZ_{n,m} := \{f \in C([0,1],M_n \otimes M_m) \mid f(0) \in M_n \otimes 1_{M_m}, f(1) \in 1_{M_n} \otimes M_m\}. 
  \end{equation}
Such an algebra is a called a prime dimension drop algebra when $n$ and $m$ are coprime. The Jiang-Su algebra $\cZ$ is the unique separable simple C*-algebra with unique tracial state which is an inductive limit of prime dimension drop algebras with unital connecting maps \cite{JiangSu99} (in fact, the dimension drop algebras can be chosen to have the form $\cZ_{n,n+1}$).It is $KK$-equivalent to $\bC$ and $\cZ$-stability is a necessary condition for $K$-theoretic classification.

By \cite[Proposition 5.1]{RordamWinter10} (or \cite[Proposition 2.1]{Sato10} for our desired formulation), $\cZ_{n,n+1}$ is the universal C*-algebra generated by elements $c_1,\dots,c_n$ and $s$ such that
\begin{itemize}
\item $c_1 \geq 0$;
\item $c_ic_j^* = \delta_{ij}c_1^2$;
\item $s^*s + \sum_{i=1}^n c_i^*c_i = 1$;
\item $c_1s = s$.
\end{itemize}
If there are uniformly tracially large (in the sense of \cite[Definition 2.2]{TomsWhiteWinter15}) order zero\footnote{order zero meaning orthogonality preserving: $\phi: A \to B$ is c.p.c. order zero if it is c.p.c. and $\phi(a)\phi(b) = 0$ whenever $ab = 0$.} c.p.c. maps $M_n \to A_\omega \cap A'$, these give rise to elements $c_1,\dots,c_n \in A_\omega \cap A'$ with $c_1 \geq 0$ and $c_ic_j^* = \delta_{ij}c_1^2$, along with certain tracial information. If $A$ has strict comparison, Matui and Sato used this tracial information to show that $A$ has property (SI) \cite{MatuiSato12}, from which one can get an element $s \in A_\omega \cap A'$ such that $s^*s + \sum_{i=1}^n c_i^*c_i = 1$ and $c_1s = s$. This gives a *-homomorphism $\cZ_{n,n+1} \to A_\omega \cap A'$, which if can be done for each $n \in \bN$, is enough to conclude that $\cZ \into A_\omega \cap A'$ unitally and hence $A \simeq A \otimes \cZ$. In fact, it suffices to show that $\cZ_{2,3} \into A_\omega \cap A'$ (or $\cZ_{n,n+1}$ for some $n \geq 2$), see \cite[Theorem 3.4(ii)]{RordamWinter10} and \cite[Theorem 5.15]{Schemaitat22}. 

\section{Approximately central approximate embeddings}\label{section:dse-localembeddings}

  Here we formalize some results on approximate embeddings. When $B \subseteq A$ is a unital inclusion of separable C*-algebras, this will yield local characterizations of nuclear subalgebras of $B_\omega \cap A'$, as defined in (\ref{eq:central-sequence-subalgebra}).

    \begin{defn}
      Let $B \subseteq A$ be a unital inclusion of C*-algebras and let $D$ be a unital, simple, nuclear C*-algebra. Let $\cF \subseteq D, \cG \subseteq A$ be finite sets and $\ee > 0$. We say that a u.c.p. map $\phi: D \to B$ is an $(\cF,\ee)$-approximate embedding if
        \begin{enumerate}
          \item $\phi(cd) \approx_\ee \phi(c)\phi(d)$ for all $c,d \in \cF$.
        \end{enumerate}
        If $\phi$ additionally satisfies
          \begin{enumerate}
            \item[2.] $[\phi(c),a] \approx_\ee 0$ for all $c \in \cF$ and $a \in \cG$,
          \end{enumerate}
          then we say that $\phi$ is an $(\cF,\ee,\cG)$-approximately central approximate embedding. 
    \end{defn}

    We will usually write that $\phi$ is a $(\cF,\ee)$-embedding or $(\cF,\ee,\cG)$-embedding to mean that $\phi$ is an $(\cF,\ee)$-approximate embedding or $(\cF,\ee,\cG)$-approximately central approximate embedding respectively. 

  \begin{remark}
    One can make a similar definition to the above if $D$ is not simple or nuclear (or even unital). The aim is to discuss subalgebras of $B_\omega \cap A'$, and if $D \into B_\omega \cap A'$ is nuclear, then one can use Choi-Effros to lift the embedding to a sequence of u.c.p. maps which are approximatey isometric, approximately multiplicative, and approximately commute with finite subsets of $A$. If $D$ is simple, the approximate isometry condition follows for free since the embedding $D \into B_\omega \cap A'$ must be isometric.
    
If we loosen the simple and nuclear assumptions on $D$, we can still speak of bounded linear maps $\phi:D \to B$ (no longer necessarily u.c.p.) which are approximately isometric, approximately multiplicative, approximately adjoint-preserving, and approximately commute with a finite prescribed subset of $A$. This will allow one to discuss general subalgebras of $B_\omega \cap A'$. As we will only be interested in strongly self-absorbing subalgebras of $B_\omega \cap A'$, which are unital, separable, simple, and nuclear (see \cite[Section 1.6]{TomsWinter07}), we restrict ourselves to u.c.p. maps from a unital, simple, nuclear C*-algebras which are approximately multiplicative and approximately commute with finite subsets of $A$.

Most of the work in this section can be done without assumptions of simplicity and nuclearity. 
  \end{remark}

    \begin{lemma}\label{lem:local-equiv-global}
      Suppose that $A,B,D$ are unital C*-algebras with $B$ separable and $D$ simple, separable and nuclear. Let $S \subseteq A$ be a separable subset. There are $(\cF,\ee,\cG)$-approximately central approximate embeddings $D \to B$ for all $\cF \subseteq D, \cG \subseteq S$ and $\ee > 0$ if and only if there is a unital embedding $D \into B_\omega \cap S'$.
        \begin{proof}
          Let $(F_n)$ be an increasing sequence of finite subsets of $D$ with dense union and let $(G_n)$ be an increasing sequence of finite subsets of $S$ with dense union. Let $\phi_n: D \to B$ be  $(F_n,\frac{1}{n},G_n)$-approximately central approximate embeddings. Let $\pi: \ell^{\infty}(B) \to B$ denote the quotient map and set
          \begin{equation}
                     \psi := \pi \circ (\phi_n): D \to B_\omega
\end{equation} 
              which is a unital embedding such that $[\psi(d),a] = 0$ for  all $d \in D$ and $a \in S$. 

          For the other direction, suppose that $\psi: D \to B_\omega \cap S'$ is a unital embedding, $\cF \subseteq D,\cG \subseteq S$ are finite and $\ee >0$. By the Choi-Effros lifting theorem (see, for example, \cite[Theorem C.3]{BrownOzawa}) there is a u.c.p. lift $\tilde{\psi} = (\tilde{\psi}_n): D \to \ell^{\infty}(B)$ such that
            \begin{itemize}
              \item $\|\tilde{\psi}_n(cd) - \tilde{\psi}_n(c)\tilde{\psi}_n(d)\| \to^{n \to \omega} 0$,
              \item $\|[\tilde{\psi}_n(d),a]\| \to^{n \to \omega} 0$ 
            \end{itemize}
            for all $c,d \in D$ and $a \in A$. Take $n$ large enough and set $\phi = \psi_n$, so that $\phi$ will be a $(\cF,\ee,\cG)$-approximately central approximate embedding. 
        \end{proof}
    \end{lemma}

      \begin{cor}
        Let $A,B,D$ be unital C*-algebras with $B,D$ separable, simple and nuclear. Suppose that there are unital embeddings $\phi: D \to B_\omega$ and $\psi: B \to A_\omega$.  Then there is a unital embedding $\xi: D \into A_\omega$. If $S \subseteq A_\omega$ is a separable subset with $\psi(B) \subseteq A_\omega \cap S'$, then $\xi$ can be chosen with $\xi(C) \subseteq A_\omega \cap S'$. 
          \begin{proof}
            Let $\cF \subseteq D$ be finite and $\ee > 0$. Let $L := \max\{\max_{d \in \cF}\|d\|,1\}$. By the above lemma, there is an $(\cF,\frac{\ee}{2L})$-approximate embedding $\phi: D \to B$, so let $\cF' = \phi(\cF)$. Now there is an $(\cF',\frac{\ee}{2L})$-approximate embedding $\psi: B \to A$. An easy calculation shows that $\psi \circ \phi: D \to A$ is an approximate $(\cF,\ee)$-embedding. 

            Appending the condition that $\psi: B \to A_\omega  \cap S'$, then, for any finite subset $\cG \subseteq S$, we can take $\psi: B \to A$ to be a $(\cF',\frac{\ee}{2L},\cG)$-approximately central approximate embedding. This gives that $\psi \circ \phi: D \to A$ is be a $(\cF,\ee,\cG)$-approximately central approximate embedding. 
          \end{proof}
      \end{cor}

        \begin{cor}
          Let  $D$ be a C*-algebra and $B \subseteq A$ be a unital inclusion of separable C*-algebras such that $B$ and $D$ are unital, separable, simple and nuclear. Suppose that there is an embedding $\pi: A \into A_\omega \cap A'$ with $\pi(B) \subseteq B_\omega \cap A'$. If $D \into B_\omega$ unitally, then $D \into B_\omega \cap A'$ unitally.
            \begin{proof}
              As $D \into B_\omega$ and $B \into B_\omega \cap A' \subseteq A_\omega \cap A'$, the above yields $D \into B_\omega \cap A'$. 
            \end{proof}
        \end{cor}


The following is useful for discussing $\cD$-stability for some inclusions of fixed point subalgebras by certain automorphisms on UHF algebras.  In particular, the following will work for automorphisms on UHF algebras of product-type, as well as tensor permutations (of finite tensor powers of UHF algebras). 

\begin{cor}\label{autfixedpoint}
  Let $A = \bigotimes_\bN B$ be an infinite tensor product of a unital, separable, nuclear C*-algebra $B$ and let $D$ be unital, separable, simple, and nuclear. Let $\lambda \in \End(A)$ be the Bernoulli shift $\lambda(a) = 1 \otimes a$. If $\sigma \in \Aut(A)$ is such that $\lambda \circ \sigma  = \sigma \circ \lambda$, and $D \into (A^\sigma)_\omega$ unitally, then $D \into (A^\sigma)_\omega \cap A'$ unitally. 
              \begin{proof}
                Note that $\pi = (\lambda^n)$ induces an embedding $A \into A_\omega \cap A'$. We just need to show that $\pi(A^\sigma) \subseteq (A^\sigma)_\omega \cap A'$. The hypothesis gives that $\lambda^n \circ \sigma = \sigma \circ \lambda^n$ for all $n$, hence $\pi(A^\sigma) \subseteq (A^\sigma)_\omega \cap A'$. The result now follows from the above. 
              \end{proof}
          \end{cor}

  We note that if we have approximately central approximate embeddings $D \to B \subseteq A$, then we can also find approximately central approximate embedding $D \to u^*Bu \subseteq A$ for any $u \in U(A)$. In the separable setting, this just means $D \into B_\omega \cap A'$ implies that $D \into u^*B_\omega u \cap A'$ for any $u \in U(A)$. 

  \begin{lemma}\label{aduaapproximateembeddings}
    Let $B \subseteq A$ be a unital inclusion of C*-algebras and let $D$ be a unital, separable, simple, nuclear C*-algebra. Let $u \in U(A)$. If there are $(\cF,\ee,\cG)$-approximately central approximate embeddings $D \to B$ for all $\cF \subseteq D, \cG \subseteq A$ finite subsets and $\ee > 0$, then there are $(\cF,\ee,\cG)$-approximately central approximate embeddings $D \to u^*Bu \subseteq A$ for all $\cF,\ee,\cG$.
      \begin{proof}
        Let $\cF \subseteq D, \cG \subseteq A$ be finite and $\ee > 0$. Let $L = \max\{1,\max_{d \in \cF}\|d\|\}$ and $\phi: D \to B$ be a $(\cF,\frac{\ee}{3L},\cG \cup \{u\})$-approximately central approximate embedding. Then $\psi = \Ad_u \circ \phi: D \to u^*Bu$ will be an $(\cF,\ee,\cG)$-embedding. 
      \end{proof}
  \end{lemma}

  We can also discuss existence of approximately central approximate embed\hyp{}dings in inductive limits (with injective connecting maps). This is an adaptation of \cite[Proposition 2.2]{TomsWinter08} to our setting. 

  \begin{prop}\label{approximateembeddinginductivelimit}
      Suppose that we have increasing sequences $(B_n)$ and $(A_n)$ of C*-algebras such that $B_n \subseteq A_n$ are unital inclusions.
      \begin{comment}
\begin{equation}
      \begin{tikzcd}
B_1 \arrow[d, hook] \arrow[r, hook] & B_2 \arrow[d, hook] \arrow[r] & \cdots  \arrow[r] & B \arrow[d, hook] \\
A_1 \arrow[r, hook]                 & A_2 \arrow[r]                 & \cdots \arrow[r]  & A,           
\end{tikzcd} 
\end{equation}
\end{comment}
If $B = \ov{\cup_n B_n}, A = \ov{\cup_n A_n}$, and $D = \ov{\cup_nD_n}$ where $(D_n)$ is an increasing sequence of unital, separable, simple, nuclear C*-algebras and there are $(\cF,\ee,\cG)$-embeddings $D_n \to B_n \subseteq A_n$ whenever $n \in \bN, \cF \subseteq D_n, \cG \subseteq A_n$ are finite and $\ee > 0$, then there are $(\cF,\ee,\cG)$-embeddings $D \to B \subseteq A$ for all $\cF \subseteq D,\cG \subseteq A$ finite and $\ee > 0$. 
  \begin{proof}
    Let $\cF \subseteq \cD$ and $\cG \subseteq A$ be finite sets and $\ee > 0$. Let
\begin{equation}
    L := \max\{1,\max_{d \in \cF}\|d\|,\max_{a \in \cG}\|a\|\}
\end{equation}
     and set $\delta := \frac{\ee}{6L+5}$. Without loss of generality assume that $\ee < 1$. Label $\cF = \{d_1,\dots,d_p\}$ and $\cG = \{a_1,\dots,a_q\}$ and find $N$ large enough so that there are $d_1',\dots,d_p' \in D_N$ and $a_1',\dots,a_q' \in A_N$ with $d_i' \approx_\delta d_i$ and $a_j' \approx_\delta a_j$. Let $\cF':=\{d_1',\dots,d_p'\}, \cG':=\{a_1',\dots,a_q'\}$ and let $\phi: D_N \to B_N \subseteq A_N$ be an $(\cF',\delta,\cG')$-embedding. As $D_N$ is nuclear, there are $k \in \bN$ and u.c.p. maps $\rho: D_N \to M_k$ and $\eta: M_k \to B_N$ such that $\eta \circ \rho(d_i') \approx_\delta \phi(d_i')$ and $\eta \circ \rho(d_i'd_j') \approx_\delta \phi(d_i'd_j')$. Use Arveson's extension theorem (see \cite[Section 1.6]{BrownOzawa}) to extend $\rho$ to a u.c.p. map $\tilde{\rho}: D \to M_k$ and let $\psi := \eta \circ \tilde{\rho}: D \to B_N$. As $B_N \subseteq B$, we can think of $\psi$ as a map $\psi: D \to B$. Now for $i=1,\dots,p$, we have
\begin{equation}
\begin{split}
        \psi(d_id_j) &\approx_{(2L+1)\delta}\psi(d_i'd_j') \\
                     &= \eta \circ \rho(d_i'd_j') \\
                     &\approx_\delta \phi(d_i'd_j') \\
                     &\approx_\delta \phi(d_i')\phi(d_j') \\
                     &\approx_{2L\delta} \eta \circ \rho(d_i')\eta \circ \rho(d_j') \\
                     &= \psi(d_i')\psi(d_j') \\
                     &\approx_{(2L+1)\delta}\psi(d_i)\psi(d_j).
\end{split}
\end{equation}
      Thus $\psi(d_id_j) \approx_{(4+6L)\delta} \psi(d_i)\psi(d_j)$, and as $(4 + 6L)\delta \leq (6L + 5)\delta = \ee$, this implies that $\psi(d_id_j) \approx_\ee \psi(d_i)\psi(d_j)$. For approximate commutation with $\cG$, we make use of the following two approximations: for $a,a',a'',b,b'$ elements in a C*-algebra,
      \begin{equation}
      \begin{split}
      &\|[a,b]\|  \leq (\|a\| + \|a'\|)\|b - b'\| + (\|b\| + \|b'\|)\|a - a'\| + \|[a',b']\|, \\
      &\|[a',b']\| \leq 2\|b'\|\|a' - a''\| + \|[a'',b']\|.
      \end{split}
      \end{equation}
        Note that for $a = \psi(d_i),a' = \psi(d_i'),a'' = \phi(d_i'), b = a_j, b' = a_j'$, we have that $\|a\|,\|b\| \leq L + 1$ and $\|a'\|,\|a''\|,\|b'\| \leq L$. Therefore from the above two inequalities we get
      \begin{equation}
      \begin{split}
      &\|[\psi(d_i),a_j]\| \leq 2L\|\psi(d_i) - \psi(c_i')\|+ 2(L+1)\|a_j - a_j'\| + \|[\psi(d_i'),a_j]\|; \\
      &\|[\psi(d_i'),a_j']\| \leq 2(L+1)\|\psi(d_i') - \phi(d_i')\| + \|[\phi(d_i'),a_j']\|.
      \end{split}
      \end{equation}
          Using these approximations we have
\begin{equation}
\begin{split}
              \|[\psi(d_i),a_j]\| &\leq 2L\|\psi(d_i) - \psi(d_i')\| + 2(L+1)\|a_j - a_j'\| + \|[\psi(d_i'),a_j]\| \\
                                &< (4L+2)\delta + \|[\psi(d_i'),a_j]\| \\
                                &\leq (4L+2)\delta + 2(L+1)\|\psi(d_i') - \phi(d_i')\| + \|[\phi(c_i'),a_j']\| \\
                                &< (4L+2)\delta + 2(L+1)\delta + \delta \\
                                &= (6L+5)\delta = \ee.
\end{split}
\end{equation}
  \end{proof}
    \end{prop}

The following will be useful to show that there are many $\cD$-stable embeddings.

\begin{lemma}\label{conjugations}
    Let $\phi: B \simeq B'$ and $\psi: A \simeq A'$ be isomorphisms between unital C*-algebras and let $D$ be a unital, simple, nuclear C*-algebra. Suppose that there is a *-homomorphism $\eta: B' \into A'$ such that there are $(\cF,\ee,\cG)$-embeddings $D \to \eta(B') \subseteq A'$ for all finite subsets $\cF \subseteq D, \cG \subseteq A'$ and $\ee > 0$. Let $\sigma = \psi^{-1} \circ \eta \circ \phi: B \to A$. Then there are $(\cF,\ee,\cG)$-embeddings $D \to \sigma(B) \subseteq A$ for all $\cF \subseteq D, \cG \subseteq A$ finite and $\ee > 0$. 
      \begin{proof}
      The diagram
      \begin{equation}
      \begin{tikzcd}
A \arrow[r, "\psi"]                      & A'                    \\
B \arrow[u, "\sigma"] \arrow[r, "\phi"'] & B' \arrow[u, "\eta"']
\end{tikzcd}
      \end{equation}
      commutes, and so if $\cF \subseteq D, \cG \subseteq A$ are finite, $\ee > 0$ and $\xi: D \to \eta(B') \subseteq A'$ is an $(\cF,\ee,\psi(\cG))$-embedding, then $\psi^{-1} \circ \xi: D \to \psi^{-1}(\eta(B')) \subseteq \psi^{-1}(A') = A$ is an $(\cF,\ee,\cG)$ embedding.  Moreover, from
      \begin{equation}
      \psi^{-1}(\eta(B')) = \psi^{-1}(\eta(\phi(B))) = \sigma(B),
      \end{equation}
      its clear that $\psi^{-1} \circ \xi$ is an $(\cF,\ee,\cG)$ embedding $D \to \sigma(B) \subseteq A$.
      \begin{comment}
        Let $\cF \subseteq D, \cG \subseteq A$ be finite and $\ee > 0$. Let $\cG' = \psi(\cG)$, and $\tilde{\xi}: D \to \eta(B')$ be an  $(\cF,\ee,\cG')$-embedding, and lift it to a u.c.p. map $\xi: D \to B$ by Choi-Effros. We claim that $\sigma \circ \phi^{-1} \circ \xi: D \to \sigma(B)$ is a $(\cF,\ee,\cG)$-embedding $D \to \sigma(B) \subseteq A$. We have
\begin{equation}
        \sigma \circ \phi^{-1} \circ \xi = \psi^{-1} \circ \eta \circ \phi \circ \phi^{-1} \circ \xi = \psi^{-1} \circ \eta \circ \xi = \psi^{-1} \circ \tilde{\xi}.
\end{equation}
 As $\xi(cd) \approx_\ee \xi(c)\xi(d)$ for $c,d \in \cF$, it is clear that $\psi^{-1} \circ \xi$ will be $(\cF,\ee)$-multiplicative, and consequently so will $\sigma \circ \phi^{-1} \circ \xi$. For approximate commutation, we have for $d \in \cF$ and $a \in \cG$:
\begin{equation}
\begin{split}
    \|[\sigma(\phi^{-1} \circ \xi(d)),a]\| &= \|[\psi^{-1} \circ \eta \circ \phi \circ \phi^{-1} \circ \xi(d),a]\| \\
                                           &= \|[\psi^{-1} \circ \eta \circ \xi(d),a]\| \\
                     &= \|[\psi \circ \psi^{-1} \circ \eta \circ \xi(d),\psi(a)]\| \\
                     &= \|[\eta \circ \xi(d),\psi(a)]\| \\
                                           &= \|[\tilde{\xi}(d),\psi(a)]\| \\
                                           &< \ee
\end{split}
\end{equation}
  where the third equality follows since $\psi$ is isometric. 
  \end{comment}
      \end{proof}
  \end{lemma}

  \section{Relative intertwinings and $\cD$-stable embeddings}\label{section:dse-embeddings}

  \subsection{Relative intertwinings}\label{ssectoin:dse-relative-intertwinings}  It is well known that a strongly self-absorbing C*-algebra $\cD$ embeds unitally into the central sequence algebra $(\cM(A))_\omega \cap A'$ of a separable C*-algebra $A$ if and only if $A \simeq A \otimes \cD$, where $\cM(A)$ is the multiplier algebra of $A$ (for example, \cite[Theorem 7.2.2(i)]{RordamBook}). We alter the proof to keep track of a subalgebra in order to show that for a unital inclusion $B \subseteq A$ of separable C*-algebras, $\cD \into B_\omega \cap A'$ if and only if there is an isomorphism $\Phi: A \to A \otimes \cD$, which is approximately unitarily equivalent to the first factor embedding, and satisfies $\Phi(B) = B \otimes \cD$. This was initially done for (irreducible) inclusions of $\II_1$ factors in \cite{Bisch90} and commented on in \cite{Izumi04} for $\cD$ being $M_{n^{\infty}},\cO_2,\cO_{\infty}$. The proof we alter is Elliott's intertwining argument, which can be found as a combination of Proposition 2.3.5, Proposition 7.2.1 and Theorem 7.2.2 of \cite{RordamBook}.

  \begin{prop}[Relative intertwining]\label{relativeintertwining}
      Let $A,B,C$ be unital, separable C*-algebras, and let $\phi: A \into C, \theta: B \to A, \psi: B \to C$ be unital *-homomorphisms such that $\phi \circ \theta(B) \subseteq \psi(B)$. Suppose there is a sequence$(u_n) \subseteq \psi(B)_\omega \cap \phi(A)'$ of unitaries such that
        \begin{itemize}
          \item $\dist(v_n^*cv_n,\phi(A)_\omega) \to 0$ for all $c \in C$;
          \item $\dist(v_n^*\psi(b)v_n,\phi \circ \theta(B)_\omega) \to 0$ for all $b \in B$.
        \end{itemize}
        Then $\phi$ is approximately unitarily equivalent to an isomorphism $\Phi: A \simeq C$ such that $\Phi \circ \theta(B) = \psi(B)$.
          \begin{proof}
            Apply the below proposition with $B_m := B, \theta_m := \theta, \psi_m := \psi$ for all $m \in \bN$.
          \end{proof}
    \end{prop}

    \begin{prop}[Countable relative intertwining]\label{countablerelativeintertwining}
        Let $A,B_m,C$ be unital, separable C*-algebras, $m \in \bN$, and $\phi: A \into C,\theta_m: B_m \to A,\psi_m: B_m \to C$ be such that $\phi \circ \theta_m(B_m) \subseteq \psi_m(B_m)$ and $\psi_1(B_1) \subseteq \psi_m(B_m)$. Suppose there is a sequence $(v_n) \subseteq \psi_1(B_1)_\omega \cap \phi(A)'$ of unitaries such that
          \begin{itemize}
            \item $\dist(v_n^*cv_n,\phi(A)_\omega) \to 0$ for all $c \in C$;
            \item $\dist(v_n^*\psi_m(b)v_n,\phi \circ \theta_m(B_m)_\omega) \to 0$ for all $b \in B_m$. 
          \end{itemize}
          Then $\phi$ is approximately unitarily equivalent to an isomorphism $\Phi:A \simeq C$ such that $\Phi \circ \theta_m (B_m) = \psi_m(B_m)$ for all $m \in \bN$. 
            \begin{proof}
              We show that if there are unitaries $(v_n) \subseteq \psi_1(B_1)$ satisfying
                \begin{itemize}
                  \item $[v_n,\phi(a)] \to 0$ for all $a \in A$;
                  \item $\dist(v_n^*cv_n,\phi(A)) \to 0$ for all $c \in C$;
                  \item $\dist(v_n^*\psi_m(b)v_n,\phi \circ \theta_m(B_m)) \to 0$ for all $b \in B_m$,
                \end{itemize}
                then the conclusion holds. Such unitaries can be found using Kirchberg's $\ee$-test (Lemma \ref{lemma:Kirchberg-eptest}).

              Let $(a_n),(b_n^{(m)}),(c_n)$ be dense sequences of $A,B_m,C$ respectively. We can inductively choose $v_n$, forming a subsequence $(v_n)$ of the unitaries above (after reindexing, we are still calling them $v_n$), such that there are $a_{jn} \in A, b_{jn}^{(m)} \in B_m$ with
                \begin{itemize}
                  \item $v_n^*\cdots v_1^*c_jv_1\cdots v_n \approx_{\frac{1}{n}} \phi(a_{jn})$;
                  \item $v_n^*\cdots v_1^*\psi(b_j^{(m)})v_1\cdots v_n \approx_{\frac{1}{n}} \phi \circ \theta_m(b_{jn}^{(m)})$;
                  \item $[v_n,\phi(a_j)] \approx_{\frac{1}{2^n}} 0$;
                  \item $[v_n,\phi(a_{jl})] \approx_{\frac{1}{2^n}} 0$;
                  \item $[v_n,\phi \circ \theta_m(b_j^{(m)})] \approx_{\frac{1}{2^n}} 0$;
                \item $[v_n,\phi \circ \theta_m(b_{jl}^{(m)})] \approx_{\frac{1}{2^n}} 0$,
                \end{itemize}
                where $j,m=1,\dots,n$ and $l=1,\dots,n-1$. Define, for $a \in \{a_n \mid n \in \bN\}$,
\begin{equation}
                \Phi(a) = \lim_n v_1\cdots v_n\phi(a) v_n^*\cdots v_1^*
\end{equation}
                which extends to a *-isomorphism $\Phi: A \simeq C$, as in \cite[Proposition 2.3.5]{RordamBook}. The proof also yields the following useful approximation:
\begin{equation}
                \Phi \circ \theta_m(b_{jn}^{(m)}) \approx_{\frac{1}{2^n}} v_1 \cdots v_n \phi \circ \theta_m(b_{jn}^{(m)})v_n^* \cdots v_1^*
\end{equation}
                for appropriate $n \geq j,m$. 

                We now need to check that $\Phi \circ \theta_m(B_m) = \psi_m(B_m)$. Approximate
\begin{equation}
                 \psi_m(b_j^{(m)}) \approx_{\frac{1}{n}} v_1\cdots v_n \phi \circ \theta_m(b_{jn}^{(m)})v_n^*\cdots v_1^* \approx_{\frac{1}{2^n}} \Phi \circ \theta_m(b_{jn}^{(m)}).
\end{equation}
                This yields $\psi_m(B_m) \subseteq \ov{\Phi \circ \theta_m(B_m)} = \Phi \circ \theta_m(B_m)$. On the other hand for any $\ee > 0$ and $b \in B_m$, we can find $n$ such that
\begin{equation}
                \Phi \circ \theta_m(b) \approx_\ee v_1\cdots v_n \phi \circ \theta_m(b)v_n^*\cdots v_1^* \in \psi_m(B_m)
\end{equation}
                since $v_i \in \psi_1(B_1) \subseteq \psi_m(B_m)$ and $\phi \circ \theta_m(B_m) \subseteq \psi_m(B_m)$. Hence $\Phi \circ \theta_m(B_m) \subseteq \ov{\psi_m(B_m)} = \psi_m(B_m)$. 
            \end{proof}
      \end{prop}

  \subsection{$\cD$-stable embeddings}\label{ssection:dse-d-stable-embeddings}

    \begin{defn}
      Let $\iota: B \into A$ be an embedding and $\cD$ be strongly self-absorbing. We say that $\iota$ is $\cD$-stable (or $\cD$-absorbing) if there exists an isomorphism $\Phi:A \simeq A \otimes \cD$ such that $\Phi \circ \iota(B) = \iota(B) \otimes \cD$. 
    \end{defn}


   We will mostly have interest in the case where $\iota$ corresponds to the inclusion map and $B \subseteq A$ is a subalgebra. In this form, we will say $B \subseteq A$ is $\cD$-stable (or $\cD$-absorbing). Clearly $\iota$ being $\cD$-stable is the same as $\iota(B) \subseteq A$ being $\cD$-stable. We note that we can define the above for any *-homomorphism. Namely, a *-homomorphism $\phi:B \to A$ is $\cD$-stable if $\phi(B) \subseteq A$ is. 



   \begin{lemma}\label{trivialinclusion}
       If $\iota: B \into A$ is an embedding, then $\iota \otimes \id_D: B \otimes \cD \into A \otimes \cD$ is $\cD$-stable.
         \begin{proof}
           Let $\phi: D \simeq D \otimes \cD$ be an isomorphism. Then
\begin{equation}
           \Phi := \id_A \otimes \phi: A \otimes \cD \to A \otimes \cD \otimes \cD
\end{equation}
           is an isomorphism with
\begin{equation}
           \Phi(\iota \otimes \id_\cD(B \otimes \cD)) = \left(\iota \otimes \id_\cD(B \otimes \cD)\right) \otimes \cD.
\end{equation}
         \end{proof}
     \end{lemma}

     We note that this is a strengthening of the notion of $\cD$-stability for C*-algebras because if $\iota = \id_A: A \to A$, then $\iota$ is $\cD$-stable if and only if $A$ is $\cD$-stable. This condition is different than the notion of $\cO_2$ or $\cO_{\infty}$-absorbing morphisms discussed in \cite{BGSW22,Gabe20,Gabe19} -- they require sequences from a larger algebra to commute with a smaller algebra, while we require sequences from a smaller algebra to commute with the larger algebra.

   The following adapts  \cite[Theorem 7.2.2]{RordamBook}.
     \begin{theorem}\label{equivDstable}
         Suppose that $B \subseteq A$ is a unital inclusion of separable C*-algebras. If $\cD$ is strongly self-absorbing, then $B \subseteq A$ is $\cD$-stable if and only if there is a unital inclusion $\cD \into B_\omega \cap A'$. 
           \begin{proof}
             Let $\phi: A \to A \otimes \cD$ be the first factor embedding $\phi(a) := a \otimes 1_\cD$. First suppose that $\xi: \cD \into B_\omega \cap A' \simeq (B \otimes 1_\cD)_\omega \cap (A \otimes 1_\cD)'$ is an embedding (so that $\phi(a)\xi(d) \in \phi(A)_\omega$ and $\phi(b)\xi(d) \in \phi(B)_\omega$). Let $\eta: \cD \into (B \otimes \cD)_\omega \cap (A \otimes 1_\cD)'$ be given by $\eta(d) := (1 \otimes d)_n$ and notice that $\xi,\eta$ have commuting ranges. As all endomorphisms of $\cD$ are approximately unitarily equivalent by \cite[Corollary 1.12]{TomsWinter07}, let $(v_n) \subseteq C^*(\xi(\cD),\eta(\cD)) \simeq \cD \otimes \cD$ be such that $v_n^*\eta(d)v_n \to \xi(d)$ for $d \in \cD$. For $b \in B$ and $d \in \cD$, we have
\begin{equation}
\begin{split}
             v_n^*(b \otimes d)v_n &= v_n^*(b \otimes 1_\cD)(1_A \otimes d)v_n^* \\
             &= v_n^*\phi(b)\eta(d)v_n \\
             &= \phi(b)v_n^*\eta(d)v_n \\
             &\to \phi(b)\xi(d) \in \phi(B)_\omega. 
             \end{split}
\end{equation}
             Moreover the same argument shows that, for $a \in A$, we have
\begin{equation}
             v_n^*(a \otimes d)v_n \to \phi(a)\xi(d) \in \phi(A)_\omega.
\end{equation}
             Now $(v_n)$ satisfy the hypothesis of Proposition \ref{relativeintertwining} with $C := A \otimes D$, $\phi$ being the first factor embedding, $\theta: B \to A$ being the inclusion and $\psi: B \simeq B \otimes \cD \subseteq A \otimes \cD = C$ (where this isomorphism exists since if $\cD \into B_\omega \cap A'$, then clearly $\cD \into B_\omega \cap B'$). From this we see that $\phi$ is approximately unitarily equivalent to an isomorphism $\Phi:A \simeq A \otimes \cD$ such that $\Phi(B) = B \otimes \cD$.

             Conversely, if $B \subseteq A$ is $\cD$-stable, let $\Phi: A \simeq A \otimes \cD$ be an isomorphism such that $\Phi(B) = B \otimes \cD$. By \cite[Proposition 1.10(iv)]{TomsWinter07}, we can identify $\cD \simeq \cD^{\otimes \infty}$ and take $\xi: \cD \into B_\omega \cap A'$ to be given by
\begin{equation}
             \xi(d) = (\Phi^{-1}(1_A \otimes 1_\cD^{\otimes n-1} \otimes d \otimes 1_\cD^{\otimes \infty}))_n.
\end{equation}
           \end{proof}
       \end{theorem}

         \begin{cor}
           Let $\iota: B \into A$ be a unital embedding between separable C*-algebras. If $\cD$ is strongly self-absorbing and $\iota$ is $\cD$-stable, then for every intermediate unital C*-algebra $C$ with $\iota(B) \subseteq C \subseteq A$, we have that $\iota(B) \subseteq C$ and $C \subseteq A$ are $\cD$-stable. In particular, $C \simeq C \otimes \cD$ for all such $C$. 
           \begin{proof}
           We have
           \begin{equation}
           \cD \into B_\omega \cap A' \subseteq B_\omega \cap C'
           \end{equation}
           and
           \begin{equation}
           \cD \into B_\omega \cap A' \subseteq C_\omega \cap A'.
           \end{equation}
           \end{proof}
         \end{cor}

         It is not however the case that any isomorphism $\Phi: A \simeq A \otimes \cD$ with $\Phi(B) = B \otimes \cD$ maps $C$ to $C \otimes \cD$.

         \begin{example}\label{noteveryintermediate}
             Let $\cD$ be strongly self-absorbing and consider
\begin{equation}
\begin{split}
                 A &:= \cD \otimes \cD \otimes \cD, \\
                 C_1 &:= \cD \otimes 1_\cD \otimes \cD,\\
                 C_2 &:= 1_\cD \otimes \cD \otimes \cD,\\
                 B &:= 1_\cD \otimes 1_\cD \otimes \cD.
\end{split}
\end{equation}
               If $f: \cD \otimes \cD \to \cD \otimes \cD$ is the tensor flip and $\phi: \cD \simeq \cD \otimes \cD$ is an isomorphism, let
\begin{equation}
                \Phi := f \otimes \phi: A \simeq A \otimes \cD
\end{equation}
               which satisfies $\Phi(B) = B \otimes \cD$ (in particular $B \subseteq A$ is $\cD$-stable). However,
\begin{equation}
               \Phi(C_1) = C_2 \otimes \cD \text{ and  } \Phi(C_2) = C_1 \otimes \cD.
\end{equation}
           \end{example}

           In fact the above example can be generalized to show that for any unital $\cD$-stable inclusion $B \subseteq A$, there is an isomorphism $\Phi: A \simeq A \otimes \cD$ with $\Phi(B) = B \otimes \cD$, and some intermediate algebra $B \subseteq C \subseteq A$ with $\Phi(C) \neq C \otimes \cD$ (obviously we will always have that $\Phi(C) \simeq C \simeq C \otimes \cD$, but equality may not happen).

             \begin{cor}
               Let $B \subseteq A$ be a $\cD$-stable inclusion. There exist a C*-algebra $C$ with $B \subseteq C \subseteq A$ and an isomorphism $\Phi: A \simeq A \otimes \cD$ such that $\Phi(B) = B \otimes \cD$ but $\Phi(C) \neq C \otimes \cD$. 
             \end{cor}


  However, we can always realize $\cD$-stability for countably many intermediate C*-algebras at once using \emph{some} isomorphism $A \simeq A \otimes \cD$.

  \begin{theorem}\label{countablymanysubalgebras}
      Suppose that $B_1 \subseteq B_m \subseteq A$ are unital inclusions of separable C*-algebras (note that we are {\bf not} asking for $(B_m)$ to form a chain). If $\cD$ is strongly self-absorbing and $\cD \into (B_1)_\omega \cap A'$ unitally, there exists an isomorphism $\Phi: A \simeq A \otimes \cD$ such that $\Phi(B_m) = B_m \otimes \cD$ for all $m \in \bN$.
        \begin{proof}
          This is essentially the same proof as Theorem \ref{equivDstable}, except we use the countable relative intertwining (Proposition \ref{countablerelativeintertwining}) in place of Proposition \ref{relativeintertwining}. Let $\xi,\eta$ be as before and let $(v_n) \subseteq C^*(\xi(\cD),\eta(\cD)) \simeq \cD \otimes \cD$ be such that $v_n^*\eta(d)v_n \to \xi(d)$ for $d \in \cD$.
            \begin{itemize}
              \item If $a \in A, d \in \cD, v_n^*(a \otimes d)v_n \to \phi(a)\xi(d) \in \phi(A)_\omega$;
              \item if $b \in B_m, v_n^*(b \otimes d)v_n \to \phi(b)\xi(d) \in \phi(B_m)_\omega$.
            \end{itemize}
            Now with  $\phi: A \to A \otimes \cD$ the first factor embedding,  $\theta_m:B_m \to A$ the inclusion maps, and $\psi_m:B_m \simeq B_m \otimes \cD$ (these exist since $\cD \into (B_1)_\omega \cap A'$ implies that $\cD \into (B_m)_\omega \cap B_m'$), our unitaries satisfy the hypothesis of Proposition \ref{countablerelativeintertwining} and therefore $\phi$ is approximately unitarily equivalent to a *-isomorphism $\Phi: A \simeq A \otimes \cD$ such that $\Phi(B_m) = B_m \otimes \cD$ for all $m$.
        \end{proof}
    \end{theorem}


    The above works since norm ultrapowers have the property that unitaries lift to sequences of unitaries.\footnote{If $u = (u_n) \in A_\omega$, then $\{n \in \bN \mid \|u_n^*u_n - 1\|\|u_nu_n^* - 1\| < 1\} \in \omega$. If $n$ is in the set, replace $u_n$ with the unitary part of its polar decomposition, and replace $u_n$ with 1 otherwise.} Tracial ultrapowers of $\II_1$ von Neumann algebras also have this property.\footnote{The tracial ultrapower of a $\II_1$ von Neumann algebra is again a $\II_1$ von Neumann algebra. Therefore if $u \in \cM^\omega$ is unitary, it is of the form $e^{ia}$ for some $a = a^* \in \cM^\omega$. Lift $a$ to a sequence $(a_n)$ of self-adjoints in $\cM$ and note that $u = (e^{ia_n})$, so that $u$ has a unitary lift.} Consequently if we work with the 2-norm $\|x\|_2 = \tau(x^*x)$ where $\tau$ is the unique trace on a $\II_1$ factor, all of the above arguments with the C*-norm replaced by $\|\cdot\|_2$ will yield back Bisch's result \cite[Theorem 3.1]{Bisch90}, provided we have the appropriate separability conditions. 

    \begin{theorem}
      Let $\cN \subseteq \cM$ be an inclusion of $\II_1$ factors with separable preduals. Then $\cR \into \cN^\omega \cap \cM'$ if and only if there exists an isomorphism $\Phi: \cM \to \cM \ov{\otimes} \cR$ such that $\Phi(\cN) = \cN \ov{\otimes} \cR$.
      \end{theorem}

  \subsection{Existence of $\cD$-stable embeddings}\label{ssection:dse-existence}

  We move to discuss the existence of $\cD$-stable embeddings. First we show that each unital embedding of unital, separable $\cD$-stable C*-algebras is approximately unitarily equivalent to a $\cD$-stable embedding. From this it will follow that there are many $\cD$-stable embeddings. 

  \begin{lemma}\label{adudstable}
      Let $\cD$ be strongly self-absorbing. If $\iota: B \into A$ is a unital, $\cD$-stable inclusion of separable C*-algebras and $u \in U(A)$, then $\Ad_u \circ \iota: B \into A$ is $\cD$-stable.
        \begin{proof}
          Apply Lemma \ref{aduaapproximateembeddings}. 
        \end{proof}
    \end{lemma}

    \begin{prop}\label{allembeddingsareue}
        Let $\cD$ be strongly self-absorbing, $A,B$ be unital separable $\cD$-stable C*-algebras and let $\iota: B \into A$ be an embedding. Then $\iota$ is approximately unitarily equivalent to a $\cD$-stable embedding $B \into A$.
          \begin{proof}
            As $A,B$ are $\cD$-stable, there are isomorphisms
\begin{equation}
             \phi: B \simeq B \otimes \cD \text{ and }\psi: A \simeq A \otimes \cD
\end{equation}
            which are approximately unitarily equivalent to the first factor embeddings $b \mapsto b \otimes 1_\cD, b \in B$ and $a \mapsto a \otimes 1_\cD, a \in A$ respectively. As $\iota \otimes \id_\cD: B \otimes \cD \into A \otimes \cD$ is $\cD$-stable by Lemma \ref{trivialinclusion},
\begin{equation}
            \sigma := \psi^{-1} \circ (\iota \otimes \id_\cD) \circ \phi: B \into A
\end{equation}
            is $\cD$-stable by Lemma \ref{conjugations}. Now we show that $\sigma$ is approximately unitarily equivalent to $\iota$. Let $\cF \subseteq B$ be finite and $\ee > 0$. Let $u \in U(B \otimes \cD)$ be such that $u^*(b \otimes 1_\cD)u \approx_\frac{\ee}{2} \phi(b)$ for $b \in \cF$ and $v \in U(A \otimes \cD)$ be such that $v^*(\iota(b) \otimes 1_\cD)v \approx_{\frac{\ee}{2}} \psi \circ \iota (b)$ for $b \in \cF$. Set $w = \psi^{-1}(\iota \otimes \id_\cD(u))^*\psi^{-1}(v) \in U(A)$. Then for $b \in \cF$,
\begin{equation}
\begin{split}
                w^*\sigma(b)w &= \psi^{-1}(v)^*\psi^{-1}(\iota \otimes \id_\cD(u\phi(b)u^*))\psi^{-1}(v) \\
                              &\approx_{\frac{\ee}{2}} \psi^{-1}(v)^*\psi^{-1}(\iota \otimes \id_\cD(b \otimes 1_\cD))\psi^{-1}(v) \\
                              &= \psi^{-1}(v)^*\psi^{-1}(\iota(b) \otimes 1_\cD)\psi^{-1}(v) \\
                              &\approx_{\frac{\ee}{2}} \psi^{-1}(\psi(\iota(b))) \\
                              &= \iota(b).
\end{split}
\end{equation}

          \end{proof}
      \end{prop}


      \begin{cor}\label{pointnormdensity}
          Let $\cD$ be strongly self-absorbing. The set of $\cD$-stable embeddings $B \into A$ of unital, separable, $\cD$-stable C*-algebras is point-norm dense in the set of embeddings $B \into A$.
            \begin{proof}
              Every embedding is approximately unitarily equivalent to a $\cD$-stable embedding. As $\cD$-stability of an embedding is preserved if one composes with $\Ad_u$, it follows that every embedding is the point-norm limit of $\cD$-stable embeddings.
            \end{proof}
        \end{cor}

        \begin{remark}\label{homomorphisminsteadofembedding}
          We note that it is not actually necessary that $\iota$ is an embedding. If $\pi: B \to A$ is any unital *-homomorphism between unital, separable, $\cD$-stable C*-algebras, then $\pi$ is approximately unitarily equivalent to a *-homomorphism $\pi': B \to A$ such that $\pi'(B) \subseteq A$ is $\cD$-stable. Consequently the set of unital *-homomorphisms $\pi: B \to A$ with $\pi(B) \subseteq A$ being $\cD$-stable is in fact dense in the set of unital *-homomorphisms $B \to A$.
        \end{remark}

        Later on, there will be some examples of non-$\cD$-stable embeddings between $\cD$-stable C*-algebras. Consequently, despite the fact $\cD$-stable embeddings are point-norm dense, the set of $\cD$-stable embeddings need not coincide with the set of all embeddings $B \into A$. Another clear consequence is that despite $\cD$-stability of an embedding being closed under conjugation by a unitary, it is not true that it is preserved under approximate unitary equivalence (in fact, the examples in question show that $\cD$-stability is not even preserved under asymptotic unitary equivalence). We finish with a corollary about embeddings into the Cuntz algebra  $\cO_2$ \cite{Cuntz77}. 

          \begin{cor}
            Let $B$ be a unital, separable, exact $\cD$-stable C*-algebra, where $\cD$ is strongly self-absorbing. Then there is a $\cD$-stable embedding $B \into \cO_2$. 
              \begin{proof}
                As $\cD$ is unital, simple, separable and nuclear by \cite[Section 1.6]{TomsWinter07}, $\cO_2 \simeq \cO_2 \otimes \cD$ and $B \into \cO_2$ unitally by Theorem 3.7 and Theorem 2.8 of \cite{KirchbergPhillips00} respectively. The above results then yield a $\cD$-stable embedding $B \into \cO_2$. 
              \end{proof}
          \end{cor}

  We include this last result about the classification of morphisms via functors. 

  \begin{theorem}
      Let $\cD$ be strongly self-absorbing and let $F$ be a functor from a class of unital, separable, $\cD$-stable C*-algebras satisfying the following.
        \begin{enumerate}
          \item[(E)] If there exists a morphism $\Phi: F(B) \to F(A)$, then there exists a *-homomorphism $\phi: B \to A$ such that $F(\phi) = \Phi$.
          \item[(U)] If $\phi,\psi: B \to A$ are *-homomorphisms which are approximately unitarily equivalent, then
\begin{equation}
             F(\phi) = F(\psi).
\end{equation}
        \end{enumerate}
        Then whenever there is a morphism $\Phi: F(B) \to F(A)$, there exists $\phi: B \to A$ such that  $F(\phi) = \Phi$ and $\phi(B)  \subseteq A$ is $\cD$-stable.
        Moreover, $\phi$ is unique up to approximate unitary equivalence.
          \begin{proof}
            By the existence (E), there exists a *-homomorphism $\phi: B \to A$. Now by Proposition \ref{allembeddingsareue} (Remark \ref{homomorphisminsteadofembedding} allows us to work with general *-homomorphisms), there exists a *-homomorphism $\phi': B \to A$ which is approx\hyp{}imately unitarily equivalent to $\phi$ and $\phi'(B) \subseteq A$ is $\cD$-stable.
            Uniqueness (U) gives that this is unique up to approximate unitary equivalence. 
          \end{proof}
    \end{theorem}

  \subsection{Permanence properties}\label{ssection:dse-permanence}
  We now discuss some permanence properties.


    \begin{lemma}
Let $\cD$ be strongly self-absorbing.
Suppose that $\iota_i: B_i \into A_i, i=1,2$ are $\cD$-stable inclusions. Then $\iota_1 \oplus \iota_2: B_1 \oplus B_2 \into A_1 \oplus A_2$ is $\cD$-stable.
        \begin{proof}
          Let $\Phi_i: A_i \simeq A_i \otimes \cD$ be isomorphisms such that $\Phi_i\circ \iota_i(B_i) = \iota_i(B_i) \otimes \cD$ and consider
\begin{equation}
          \Phi: A_1 \oplus A_2 \simeq (A_1 \oplus A_2) \otimes \cD
\end{equation}
          given by the composition
\begin{equation}
          \begin{tikzcd}
A_1 \oplus A_2 \arrow[r, "\Phi_1 \oplus \Phi_2"] & (A_1 \otimes \cD) \oplus (A_2 \otimes \cD) \arrow[r, "\simeq"] & (A_1 \oplus A_2) \otimes \cD
\end{tikzcd}
\end{equation}
where the last isomorphism follows from (finite) distributivity of the min-tensor. Then we see that
\begin{equation}
\Phi(\iota_1(B_1) \oplus \iota_2(B_2)) = \left(\iota_1(B_1) \oplus \iota_2(B_2)\right) \otimes \cD.
\end{equation}
        \end{proof}
    \end{lemma}

      \begin{lemma}
Let $\cD$ be strongly self-absorbing.        
        Suppose that $\iota_i: B_i \into A_i, i=1,2$ are inclusions and that at least one of $\iota_1$ or $\iota_2$ is $\cD$-stable. Then $\iota_1 \otimes \iota_2: B_1 \otimes B_2 \into A_1 \otimes A_2$ is $\cD$-stable. 
          \begin{proof}
            We prove this if $\iota_2$ is $\cD$-stable, and a symmetric argument will yield the result if $\iota_1$ is. Let $\Phi_2: A_2 \simeq A_2 \otimes \cD$ be such that $\Phi_2 \circ \iota_2(B_2) = \iota(B_2) \otimes \cD$. Taking
\begin{equation}
            \Phi := \id_{A_1} \otimes \Phi_2: A_1 \otimes A_2 \simeq A_1 \otimes A_2 \otimes \cD,
\end{equation}
            we have that
\begin{equation}
            \Phi(\iota_1(B_1) \otimes \iota_2(B_2)) = \iota_1(B_1) \otimes \iota_2(B_2) \otimes \cD.
\end{equation}
          \end{proof}
      \end{lemma}
      
        \begin{prop}
          Let $\cD$ be strongly self-absorbing.
Suppose that we have increasing sequences of unital separable C*-algebras $(B_n)$ and $(A_n)$ such that $B_n \subseteq A_n$ unitally.
          \begin{comment}
\begin{equation}
           \begin{tikzcd}
B_1 \arrow[d, hook] \arrow[r, hook] & B_2 \arrow[d, hook] \arrow[r] & \cdots  \arrow[r] & B \arrow[d, hook] \\
A_1 \arrow[r, hook]                 & A_2 \arrow[r]                 & \cdots \arrow[r]  & A    ,        
\end{tikzcd}
\end{equation}
\end{comment}
Let $B = \ov{\cup_nB_n}$ and $A = \ov{\cup_n A_n}$. If $B_n \subseteq A_n$ is $\cD$-stable for all $n$, then $B \subseteq A$ is $\cD$-stable.
  \begin{proof}
    This follows from Proposition \ref{approximateembeddinginductivelimit}, together with Lemma \ref{lem:local-equiv-global} and Theorem \ref{equivDstable}.
  \end{proof}
        \end{prop}


        Lastly we'll discuss unital inclusions $B \subseteq A$ of $C(X)$ algebras, where $X$ is a compact Hausdorff space. We show that if $X$ has finite covering dimension, then such an inclusion is $\cD$-stable if and only if the inclusion $B_x \subseteq A_x$ along each fibres is $\cD$-stable. 

        \begin{lemma}\label{imageofinclusion}
            Let $\cD$ be strongly self-absorbing.
Suppose that $B_i \subseteq A_i$ are unital inclusions, for $i=1,2$, and $\psi: A_1 \to A_2$ is a surjective *-homomorphism such that $\psi(B_1) = B_2$. If $B_1 \subseteq A_1$ is $\cD$-stable, then so is $B_2 \subseteq A_2$.
              \begin{proof}
                We note that $\psi$ induces a *-homomorphism
\begin{equation}
                \tilde{\psi}: (B_1)_\omega \cap A_1' \to (B_2)_\omega \cap A_2'
\end{equation}
                and consequently if $\xi: \cD \into (B_1)_\omega \cap A_1'$, we have a unital *-homomorphism
\begin{equation}
                \eta := \tilde{\psi} \circ \xi: \cD \to (B_2)_\omega \cap A_2'.
\end{equation}
                $\eta$ is automatically injective since $\cD$ is simple. 
              \end{proof}
          \end{lemma}


Rephrasing the above in terms of commutative diagrams, it says that if we have a commutative diagram
\begin{equation}
\begin{tikzcd}
A_1 \arrow[r, two heads]                 & A_2                 \\
B_1 \arrow[r, two heads] \arrow[u, hook] & B_2 \arrow[u, hook]
\end{tikzcd}
\end{equation}
where the left inclusion is $\cD$-stable, then the right inclusion is $\cD$-stable as well. 

Now we consider many of the results discussed in \cite[Section 4]{HirshbergRordamWinter07}, except for inclusions of C*-algebras.

\begin{defn}
      Let $X$ be a compact Hausdorff space. A $C(X)$-algebra is a C*-algebra $A$ endowed with a unital *-homomorphism $C(X) \to \cZ(\cM(A))$, where $\cZ(\cM(A)$ is the center of the multiplier algebra $\cM(A)$ of $A$. 
    \end{defn}

    If $Y \subseteq X$ is a closed subset, we set $I_Y := C_0(X \sm Y)A$, which is a closed two-sided ideal in $A$. We denote $A_Y := A/A_Y$ and the quotient map  $A \to A_Y$ by $\pi_Y$. For an element $a \in A$, we write $a_Y := \pi_Y(a)$ and if $Y$ consists of a single point $x$, we write $A_x,I_x,\pi_x$ and $a_x$. We say that $A_x$ is the fibre of $A$ at $x$. We note that $A_X = A$.
    
    If $B \subseteq A$ is a unital inclusion and $\theta:_A: C(X) \to A, \theta_B: C(X) \to B$ are morphisms witness $A$ and $B$ as $C(X)$-algebras, respectively, we say that $B \subseteq A$ is an inclusion of $C(X)$-algebras if
    \begin{equation}
    \begin{tikzcd}
B \arrow[r, hook]                                  & A \\
C(X) \arrow[u, "\theta_B"] \arrow[ru, "\theta_A"'] &  
\end{tikzcd}
    \end{equation}
    commutes. Note that $\theta_B(B) \subseteq \cZ(A)$. We note that when discussion an inclusion of fibres $B_Y \subseteq A_Y$ we are considering $B_Y := \pi^A_Y(B) \subseteq \pi^A_Y(A) =: A_Y$, where $\pi_Y^A: A \to A_Y$ is the associated quotient map. 
   
    
  \begin{remark}[Upper semi-continuity]\label{rem:up-semi-cts}
  In \cite[Section 1.3]{HirshbergRordamWinter07}, it was pointed out that the norm on a $C(X)$-algebra $A$ is upper semi-continuous. This meaning that, fixing some $a \in A$, the function $x \mapsto \|a_x\|$ from $X$ to $\bR$ is upper semi-continuous (as it is the infimum of a family of continuous functions), and consequently the set $\{x \in X \mid \|a_x\| < \ee\} \subseteq X$ is open for all $a \in A$ and $\ee > 0$. 
  \end{remark}
  
 
 \begin{comment}
  For two unital $C(X)$-algebras $A,B$, let $\theta_A: C(X) \to Z(A)$ and $\theta_B: C(X) \to Z(B)$ be the unital *-homomorphisms in the above definition. We say that a *-homomorphism $\phi: B \to A$ is a $C(X)$-morphism if
\begin{equation}
  \phi \circ \theta_B = \theta_A. 
\end{equation}
  Note that for a closed subset $Y \subseteq X$, a $C(X)$-morphism induces a *-homomorphism $\phi_Y: B_Y \to A_Y$ given by
\begin{equation}
  \phi_Y(\pi_Y(b)) = \pi_Y(\phi (b)).
\end{equation}
 This gives a morphism of short exact sequences:
\begin{equation}
 \begin{tikzcd}
0 \arrow[r] & I_Y^{B} \arrow[r] \arrow[d, "\phi|_{I_Y^{B}}"] & B \arrow[r, "\pi_Y^{B}"] \arrow[d, "\phi"] & B_Y \arrow[r] \arrow[d, "\phi_Y"] & 0 \\
0 \arrow[r] & I_Y^{A} \arrow[r]                              & A \arrow[r, "\pi_Y^A"']                    & A_Y \arrow[r]                     & 0.
\end{tikzcd}
\end{equation}
%\begin{comment}
Given an embedding $\phi: B \to A$ of $C(X)$-algebras, we will be interested in the fibrewise inclusions $\phi_Y(B_Y) \subseteq A_Y$.
\end{comment}

We note that Lemma \ref{imageofinclusion} gives that if $B\subseteq A$ is $\cD$-stable and $Y \subseteq X$ is closed, then $B_Y \subseteq A_Y$ is automatically $\cD$-stable as well since we have the commuting diagram
\begin{equation}
\begin{tikzcd}
A \arrow[r, "\pi_Y"]                     & A_Y                \\
B \arrow[u, hook] \arrow[r, "\pi_Y|_B"'] & B_Y. \arrow[u, hook]
\end{tikzcd}
\end{equation}
The converse needs a bit of work. This is the embedding-analogue of the beginning of \cite[Section 4]{HirshbergRordamWinter07}. We discuss how the proofs can be adapted and often omit approximations that were otherwise done there. We want a version of \cite[Lemma 4.5]{HirshbergRordamWinter07}, which is a result about \emph{gluing} c.c.p. maps together along fibres. In our setting, we are only interested in u.c.p. maps, and we want to show that if we \emph{glue} two u.c.p. maps together whose images are contained in some $C(X)$-subaglebra $B$, then the \emph{glued} map also has image contained in $B$. We borrow their Definition 4.2.

\begin{defn}
Let $A$ be a unital $C(X)$-algebra, for a compact Hausdorff space $X$, and let $D$ be a unital C*-algebra. Let $\phi: D \to A$ is a u.c.p. map and $Y \subseteq X$ a closed subset. If $\cF \subseteq D, \cG \subseteq A$ are finite and $\ee > 0$, we say that $\phi$ is $(\cF,\ee,\cG)$-good for $Y$ if
\begin{enumerate}
\item $([\phi(d),a])_Y \approx_\ee 0$ and
\item $\phi(dd')_Y \approx_\ee \phi(d)_Y\phi(d')_Y$
\end{enumerate}
whenever $d,d' \in \cF$ and $a \in \cG$. If $X = [0,1]$, $Y \subseteq X$ is a closed interval, $\cF' \supseteq \cF$ is another finite set and $0 < \ee' < \ee$, we say that $\phi$ is $(\cF,\ee,\cG;\cF',\ee')$-good for $Y$ if $\phi$ is $(\cF,\ee,\cG)$-good for $Y$ and there exists some closed neighbourhood $V$ of the endpoints of $Y$ such that $\phi$ is $(\cF',\ee',\cG)$-good for $V$. 
\end{defn}

First we need a lemma that follows as a consequnce of $\cD$-stability. It is the embedding analogue of \cite[Proposition 4.1]{HirshbergRordamWinter07}.

\begin{lemma}\label{lem:kappa-mu}
Let $\cD$ be strongly self-absorbing, and $B \subseteq A$ be a unital, $\cD$-stable inclusion of separable C*-algebras. Then for any $\cG \subseteq A$ finite and $\ee > 0$, there exist unital *-homomorphisms $\kappa: A \to A$ and $\mu: \cD \to B$ such that
\begin{enumerate}
\item $\kappa(B) \subseteq B$,
\item $[\kappa(A),\mu(\cD)] = 0$,
\item $\kappa(a) \approx_\ee a$ for all $a \in \cG$.
\end{enumerate}
\begin{proof}
The proof is essentially the same as the proof of (a) $\Rightarrow (c)$ in \cite[Proposition 4.1]{HirshbergRordamWinter07}. As $B \subseteq A$ is $\cD$-stable, let us identify $B \subseteq A$ with $B \otimes \cD \subseteq A \otimes \cD$. As $\cD$ is strongly self-absorbing,  \cite[Theorem 2.3]{TomsWinter07} gives a sequence $(\phi_n)$ of *-homomorphisms $\phi_n: \cD \otimes \cD \to \cD$ such that
\begin{equation}
\phi_n(d \otimes 1_\cD) \to d \text{ for all } d \in \cD.
\end{equation}
Define $\kappa_n: A \otimes \cD \to A \otimes \cD$ by
\begin{equation}
\kappa_n:= (\id_A \otimes \phi) \circ (\id_A \otimes \id_\cD \otimes 1_\cD), 
\end{equation}
and $\mu_n: \cD \to B \otimes \cD$ by
\begin{equation}
\mu_n := (\id_B \otimes \phi_n) \circ (1_A \otimes 1_\cD \otimes \id_\cD).
\end{equation}
Then taking $n$ large enough and letting $\kappa$ and $\mu$ be $\kappa_n$ and $\mu_n$ respectively, its clear that $\kappa(B \otimes \cD) \subseteq B \otimes \cD$, $[\kappa(A),\mu(\cD)] = 0$ and that $\kappa(a) \approx_\ee a$ whenever $a$ is in some prescribed finite subset $\cG \subseteq A$ and $\ee > 0$ is some prescribed error. 
\end{proof}
\end{lemma}

\begin{lemma}
Let $\cD$ be strongly self-absorbing and $A$ be a unital, separable $C([0,1])$-algebra. Suppose $\cF \subseteq \cD, \cG \subseteq A$ are finite self-adjoint subsets of contractions with $1_\cD \in \cF$. Suppose that we have points $0 \leq r < s < t \leq 1$ and two u.c.p. maps $\rho,\sigma: \cD \to A$ which are $(\cF,\ee,\cG)$-good for $[r,s],[s,t]$ respectively. Suppose that $A_s$ is $\cD$-stable. 

Then there are u.c.p. maps $\rho',\sigma': \cD \to A$ which are $(\cF,\ee,\cG)$-good for $[r,s],[s,t]$ respectively, and u.c.p. maps $\nu_{\rho'},\nu_{\sigma'}: \cD \to A,\mu_{\rho'},\mu_{\sigma'}: \cD \otimes \cD \to A$ such that $\nu_{\rho'},\nu_{\sigma'}$ are $(\cF,3\ee,\cG)$-good for some interval $I \subseteq (r,t)$ containing $s$ in its interior, and such that for any $a \in \cG, d,d' \in \cF$, we have
\begin{enumerate}
\item $([\rho'(d),\nu_{\rho'}(d')])_I \approx_{2\ee} 0$
\item $([\sigma'(d),\nu_{\sigma'}(d')])_I \approx_{2\ee} 0$
\item $\rho'(d)_I\nu_{\rho'}(d')_I \approx_\ee \mu_{\rho'}(d \otimes d')_I$
\item $\sigma'(d)_I\nu_{\sigma'}(d')_I \approx_\ee \mu_{\sigma'}(d \otimes d')_I$
\item $\nu_{\rho'}(d)_I \approx_{2\ee} \nu_{\sigma'}(d)_I$.
\end{enumerate} 
If $\rho,\sigma$ are $(\cF,\ee,\cG;\cF',\ee)$-good for $[r,s],[s,t]$ respectively, for some finite $\cF' \supseteq \cF$ set of contractions and for some $0 < \ee' < \ee$, then we can arrange so that $\rho',\sigma',\nu_{\rho'},\nu_{\sigma'}$ are $(\cF',3\ee',\cG)$-good for the interval $I$, and that the above five conditions hold with $\ee'$ in place of $\ee$ and $\cF'$ in place of $\cF$. 

Moreover, if $B \subseteq A$ is a unital inclusion of $C([0,1])$-algebras such that $\rho(\cD) \subseteq B, \sigma(\cD) \subseteq B$ and $B_s \subseteq A_s$ is $\cD$-stable, then the images of all $\rho',\sigma',\mu_{\rho'},\mu_{\sigma'}$ are contained in $B$ (as are the images of $\nu_{\rho'}$ and $\nu_{\sigma'}$).
\begin{proof}
This is \cite[Lemma 4.4]{HirshbergRordamWinter07}, except we've replaced c.c.p. maps with u.c.p. maps. One can easily check that the resulting maps are u.c.p. maps. 

As for the ``moreover'' part, which is the only addition besides the unitality, we outline the definitions of these maps to show that the images of $\rho',\sigma',\mu_{\rho'},\mu_{\sigma'}$ are contained in $B$. As $B_s \subseteq A_s$ is $\cD$-stable, we can find $\kappa: A_s \to A_s$ and $\mu: \cD \to B_s$ as in Lemma \ref{lem:kappa-mu}, where $\kappa(a_s) \approx a_s$ for an appropriate error whenever $a \in \cG$. We use Choi-Effros to find u.c.p. lifts $\tilde{\rho},\tilde{\sigma}: \cD \to B$ for the maps $\kappa \circ \pi_s \circ \rho$ and $\kappa \circ \pi_s \circ \sigma$ respectively (note that $\kappa \circ \pi_s \circ \rho$ and $\kappa \circ \pi_s \circ \sigma$ lie in $B_s$, which is a *-homomorphism image of $B$). One then defines piece-wise linear functions $f,g: [0,1] \to [0,1]$ which attain both values 0 and 1 at the end points (their definition is not important to show the ``moreover'' part). Then $\rho',\sigma'$ are defined as
\begin{equation}
\rho'(d) := (1-f)\cdot \rho(d)  + f\cdot\tilde{\rho}(d) \text{ and } \sigma'(d) := (1-g)\cdot\sigma(d) + g\cdot \tilde{\sigma}(d)
\end{equation}
Clearly $\rho',\sigma'$ take values in $B$ as $\rho,\tilde{\rho},\sigma,\tilde{\sigma}$ all do and $(1-f),f,(1-g),g$ are in $B$. Now we define u.c.p. maps $\tilde{\mu}_{\rho'},\tilde{\mu}_{\sigma'}: \cD \otimes \cD \to B_s$ by
\begin{equation}
\tilde{\mu}_{\rho'}(d \otimes d') := \rho'(d)_s\mu(d') \text{ and }\tilde{\mu}_{\sigma'}(d \otimes d') := \sigma'(d)_s\mu(d').
\end{equation}
Now by Choi-Effros, we can take u.c.p. lifts $\mu_{\rho'}$ and $\mu_{\sigma'}$ of $\tilde{\mu}_{\rho'}$ and $\mu_{\sigma'}$, respectively. As the images of $\tilde{\mu}_{\rho'}$ and $\mu_{\sigma'}$ lie in $B_s$, the images of $\mu_{\rho'}$ and $\mu_{\sigma'}$ will lie in $B$. 
\end{proof}
\end{lemma}

\begin{lemma}\label{lem:exist-F-ee}
Let $A$ be a unital, separable $C([0,1])$-algebra. Suppose $\cF \subseteq \cD, \cG \subseteq A$ are finite self-adjoint subsets with $1_\cD \in \cF$ and $\ee > 0$. There exists $0 < \ee' < \ee$ and a finite subset $\cF' \supseteq \cF$ such that if $\rho,\sigma: \cD \to A$ are u.p.c. maps and $0 \leq r < s < t \leq 1$ are points such that $\rho$ is $(\cF,\ee,\cG;\cF',\ee')$-good for $[r,s]$,  $\sigma$ is $(\cF,\ee,\cG;\cF',\ee')$-good for $[s,t]$ and $A_s$ is $\cD$-stable, then there is a u.c.p. map $\psi: \cD \to A$ which is $(\cF,\ee,\cG;\cF',\ee')$-good for $[r,t]$. 

Moreover, if $B \subseteq A$ is a unital inclusion of $C([0,1])$-algebras such that $\rho(\cD) \subseteq B$, $\sigma(\cD) \subseteq B$ and $B_s \subseteq A_s$ is $\cD$-stable, then $\psi(\cD) \subseteq B$.
\begin{proof}
The first part is \cite[Lemma 4.5]{HirshbergRordamWinter07}, except we've replaced c.c.p. maps with u.c.p. maps. One has to check that the resulting $\psi$ is unital, but this follows easily if $\rho$ and $\sigma$ are.

We outline the construction of $\psi$ to show unitality, as it will also be useful to show the ``moreover'' part, which is the only real addition. Let $u \in C([0,1],\cD \otimes \cD)$ be a path of unitaries such that $u_0 = 1_{\cD \otimes \cD}$ and
\begin{equation}
u_1(d \otimes 1_\cD)u_1^* \approx_{\frac{\ee}{4}} 1_\cD \otimes d. 
\end{equation}
We replace $\rho,\sigma$ with $\rho',\sigma'$ as in the above lemma and this yields u.c.p. maps $\mu_\rho,\mu_\sigma$ satisfying the hypotheses above for some interval $I \subseteq (s,t)$ with $s$ in its interior. Define
\begin{equation}
\phi_\rho,\phi_\sigma: C([0,1]) \otimes \cD \otimes \cD \to A
\end{equation}
by
\begin{equation}
\begin{split}
\phi_\rho(f \otimes d \otimes d') &:= f\cdot \mu_\rho(d \otimes d') \\
\phi_\sigma(f \otimes d \otimes d') &:= f\cdot \mu_\sigma(d \otimes d'). 
\end{split}
\end{equation}
Note that these maps are unital. Take piece-wise linear functions $h_1,h_2,h_3,h_4: [0,1] \to [0,1]$ which sum to 1 (their specific form does not matter to show unitality of $\psi$ nor the ``moreover'' part) and  $g_\rho,g_\sigma: [0,1] \to [0,1]$ which sum to 1 (again, their specific form does not matter to show unitality of $\psi$ nor the ``moreover'' part). Define unitaries $u_\rho,u_\sigma \in C([0,1]) \otimes \cD \otimes \cD \simeq C([0,1],\cD \otimes \cD)$ by 
\begin{equation}
u_{\rho x} := u_{g_\rho(x)} \text{ and } u_{\sigma x} := u_{g_\sigma(x)}.
\end{equation}
Now define $\zeta_\rho,\zeta_\sigma: \cD \to A$ by
\begin{equation}
\begin{split}
\zeta_\rho(d) &:=  \phi_\rho(u_\rho(1_{C([0,1])} \otimes d \otimes 1_\cD)u_\rho^*) \\
\zeta_\sigma(d) &:= \phi_\sigma(u_\sigma(1_{C([0,1])} \otimes d \otimes 1_\cD)u_\sigma^*),
\end{split}
\end{equation}
which are clearly unital. Finally the map $\psi: \cD \to A$ is defined by
\begin{equation}
\psi(d) := h_1\cdot\rho(d) + h_2\cdot \zeta_\rho(d) + h_3\cdot \zeta_\sigma(d) + h_4\cdot \sigma(d).
\end{equation}
Clearly $\psi$ is unital. 

Now for the ``moreover'' part. If $\rho(\cD) \subseteq B$ and $\sigma(\cD) \subseteq B$, clearly the first and fourth terms in the definition of $\psi$ will lie in $B$. So it suffices to show that $\zeta_\rho(\cD) \subseteq B$ and $\zeta_\sigma(\cD) \subseteq B$, and for this it suffices to show that $\mu_\rho(\cD \otimes \cD) \subseteq B$ and $\mu_\sigma(\cD \otimes \cD) \subseteq B$ (since $h_1,h_2,h_3,h_4$ all lie in $B$). But this follows from the ``moreover'' part of the previous lemma. 
\end{proof}
\end{lemma}

  With this, we get the analogue of their Theorem 4.6, the proof being essen\hyp{}tially the same as well, except we insist that the our u.c.p. maps commute with a prescribed finite subset of $A$. 
  
\begin{prop}
Let $\cD$ be strongly self-absorbing, and $X$ be a compact Hausdorff space with finite covering dimension. Suppose that $B \subseteq A$ is a unital inclusion of $C(X)$-algebras. Then $B_x \subseteq A_x$ is $\cD$-stable for all $x \in X$ if and only if $B \subseteq A$ is $\cD$-stable. 
          \begin{proof}
            As previously mentioned, if $B \subseteq A$ is $\cD$-stable, then $B_x \subseteq A_x$ is $\cD$-stable for all $x$.

            For the converse, the proof is essentially the same as \cite[Theorem 4.6]{HirshbergRordamWinter07}. Using the arguments there, one can simplify to the case where we can argue this for $C([0,1])$-algebras (by using \cite[Theorem V.3]{HurewiczWallman}, which says that a compact space of dimension $\leq n$ is homeomorphic to a subset of $[0,1]^{2n+1}$, and then working component-wise). Now for $\cF \subseteq \cD, \cG \subseteq A$ and  $\ee > 0$, let $\cG_x := \{a_x \mid a \in \cG\}$. Without loss of generality suppose that $\cF^*=\cF,\cG^*=\cG$ and that $1_\cD \in \cF$. Let $\cF',\ee'$ be as in Lemma \ref{lem:exist-F-ee}. 
            
            By $\cD$-stability of the inclusion $B_x \subseteq A_x$ there are u.c.p. $(\cF',\ee',\cG_x)$-embed\hyp{}dings $\psi_x: \cD \to B_x \subseteq A_x$ which lift by Choi-Effros to u.c.p. maps $\psi_x': \cD \to B$. The norm is upper semi-continuous (Remark \ref{rem:up-semi-cts}), and this yields intervals $I_x \subseteq [0,1]$ such that $\psi_x'$ is $(\cF',\ee',\cG)$-good for $\ov{I_x}$.  Note that $\psi_x'$ being $(\cF',\ee',\cG)$-good for the whole of $I_x$ implies that it is $(\cF,\ee,\cG;\cF',\ee')$-good for $\ov{I_x}$.  
             Compactness then allows us to split the interval as
\begin{equation}
            0 = t_0 < t_1 < \dots < t_n = 1
\end{equation}
            and to take $\psi_i: \cD \to B$ u.c.p.  which are $(\cF,\ee,\cG;\cF',\ee')$-good for $[t_{i-1},t_i]$ for $i=1,\dots,n$ ($\psi_i = \psi_x'$ for some $x \in [0,1]$). Now by repeatedly using the gluing lemma (Lemma \ref{lem:exist-F-ee}) to glue these maps together, we can find a u.c.p. map $\psi: \cD \to B$ which is an $(\cF,\ee,\cG)$-embedding. 
          \end{proof}
\end{prop}


  \section{Crossed products}\label{section:dse-crossed-products}

  In this section we discuss how inclusions coming from non-commutative dynamics fit into the framework of a tensorially absorbing inclusions. We'll shortly discuss group actions $G \act^\alpha A$ with Rokhlin properties and consider the inclusion of a C*-algebra in its crossed product $A \subseteq A \rtimes_\alpha G$, as well as the inclusion of the fixed point subalgebra of the action in the C*-algebra $A^\alpha \subseteq A$. We then discuss diagonal inclusions associated to certain group actions. 

  This first result says that if we have an isomorphism $A \simeq A \otimes \cD$ which is $G$-equivariant with respect to an action point-wise fixing the right tensor factor, then the corresponding inclusion $A \subseteq A \rtimes_{r,\alpha} G$ is $\cD$-stable.  

  \begin{prop}
    Let $G \act^\alpha A$ be an action of a countable discrete group on a unital separable C*-algebra. Suppose that $\alpha \simeq \alpha \otimes \id_\cD$, that is, there is an isomorphism $\Phi: A \simeq A \otimes \cD$ such that
\begin{equation}
 \begin{tikzcd}
A \arrow[d, "\alpha_g"'] \arrow[r, "\Phi"] & A \otimes \cD \arrow[d, "\alpha_g \otimes \id_\cD"] \\
A \arrow[r, "\Phi"']                       & A \otimes \cD                                      
\end{tikzcd}
\end{equation}
commutes for all $g \in G$. Then $A \subseteq A \rtimes_{r,\alpha} G$ is $\cD$-stable. 
      \begin{proof}
          Let $\psi: \cD \simeq \cD^{\otimes \infty}$ and let $\phi_n: \cD \to \cD^{\otimes \infty}$ be the $n$th factor embedding:
\begin{equation}
            \phi_n(d)  := 1_\cD^{\otimes n-1} \otimes d \otimes 1_\cD^{\otimes \infty}.
\end{equation}
            We claim that $\xi(d) := (\Phi^{-1}(1_A \otimes \psi^{-1} \circ \phi_n(d)))_n: \cD \to A_\omega$ is an embedding such that $\xi(\cD) \subseteq A_\omega \cap A'$ and $(\alpha_g)_\omega \circ \xi = \xi$ for all $g \in G$ -- that is, $\xi$ is an embedding $\cD \into A_\omega \cap (A \rtimes_{r,\alpha} G)'$. The first claim is obvious, so we prove the second. We have
\begin{equation}
\begin{split}
                \|\alpha_g(\Phi^{-1}(1_A &\otimes \psi^{-1}(\phi_n(d)))) -  \Phi^{-1}(1_A \otimes \psi^{-1}(\phi_n(d)))\| \\
                                         &= \|\Phi \circ \alpha_g(\Phi^{-1}(1_A \otimes \psi^{-1}(\phi_n(d)))) -  \Phi(\Phi^{-1}(1_A \otimes \psi^{-1}(\phi_n(d))))\| \\
                                         &= \|\alpha_g \otimes \id_\cD(1_A \otimes \psi^{-1}(\phi_n(d))) - 1_A \otimes \psi^{-1}(\phi_n(d)))\| \\
                                         &=0. 
\end{split}
\end{equation}
      \end{proof}
  \end{prop}

  The next lemma of note is the following. 

    \begin{lemma}\label{lem:fixed-point-in-crossed-product-D-stable}
      Suppose that $G \act^\alpha A$ is an action of a finite group on a unital separable C*-algebra $A$ such that $A \subseteq A \rtimes_\alpha G$ is $\cD$-stable. Then $A^\alpha \subseteq A \rtimes_\alpha G$ is $\cD$-stable. In particular, if $A \subseteq A\rtimes_\alpha G$ is $\cD$-stable, then $C \simeq C \otimes \cD$ whenever $A^\alpha \subseteq C \subseteq A \rtimes_\alpha \cD$.
        \begin{proof}
        For an element $(x_n) \in A_\omega \cap (A \rtimes_\alpha G)'$, an easy averaging argument shows that
\begin{equation}
        (x_n) = \left(\frac{1}{|G|}\sum_{g \in G} \alpha_g(x_n)\right)
\end{equation}
in $A_\omega$, and the right is clearly point-wise fixed by $\alpha_g$ for all $g \in G$. So $A_\omega \cap (A \rtimes_\alpha G)'$ is actually equal to $(A^\alpha)_\omega \cap (A \rtimes_\alpha G)'$, and the existence of a unital embedding of $\cD$ in $A_\omega \cap (A \rtimes_\alpha G)'$ is in fact equivalent to the existence of a unital embedding of $\cD$ into $(A^\alpha)_\omega \cap (A \rtimes_\alpha G)'$. The result follows.
        \end{proof}
    \end{lemma}


    Using the Galois correspondence of Izumi \cite{Izumi02} yields the following.

    \begin{theorem}\label{Galois}
        Let $A$ be a unital, simple, separable C*-algebra and let $G \act^\alpha A$ be an action of a finite group by outer automorphisms. If $A \subseteq A \rtimes_\alpha \cD$ is $\cD$-stable, then there exists an isomorphism $\Phi: A \rtimes_\alpha G \simeq (A \rtimes_\alpha G) \otimes \cD$ such that whenever $C$ is a unital C*-algebra satisfying either
          \begin{enumerate}
            \item $A^\alpha \subseteq C \subseteq A$ or
            \item $A \subseteq C \subseteq A \rtimes_\alpha G$,
          \end{enumerate}
          we have $\Phi(C) = C \otimes \cD$.
            \begin{proof}
              Applying \cite[Corollary 6.6]{Izumi02} gives the following two correspondences:
                \begin{enumerate}
                  \item there is a one-to-one correspondence between subgroups of $G$ with intermediate C*-algebras $A^\alpha \subseteq C \subseteq A$ given by
\begin{equation}
                    H \leftrightarrow A^{\alpha_H};
\end{equation}
                  \item there is a one-to-one correspondence between subgroups of $G$ and intermediate C*-algebras $A \subseteq C \subseteq A \rtimes_\alpha G$ given by
\begin{equation}
                    H \leftrightarrow A \rtimes_{\alpha|_H} H.
\end{equation}
                \end{enumerate}
            In particular, there are only finitely many C*-algebras $C$ between either $A^\alpha \subseteq A$ or $A \subseteq A \rtimes_\alpha G$. As all such lie between the $\cD$-stable inclusion $A^ \alpha \subseteq A \rtimes G$, Theorem \ref{countablymanysubalgebras} yields the desired isomorphism. 
            \end{proof}
      \end{theorem}



  \subsection{(Tracial) Rokhlin properties}\label{ssection:dse-tracial-Rokhlin}

  Here we will restrict ourselves to finite groups for simplicity, although many results hold more generally (see \cite{HirshbergWinter07,HirshbergOrovitz13,GardellaHirshberg18}). 

  \begin{defn}
      Let $A$ be a unital, separable C*-algebra. We say that a finite group action $G \act^\alpha A$ has the Rokhlin property if there are pairwise orthogonal projections $(p_g)_{g \in G} \subseteq A_\omega \cap A'$ summing to $1_{A_\omega}$ such that $(\alpha_g)_\omega(p_h) = p_{gh}$ for $g,h \in G$.
    \end{defn}

    \begin{prop}
      Let $A$ be a unital, separable $\cD$-stable C*-algebra. If $G \act^\alpha A$ is an action of a finite group with the Rokhlin property, then $A^\alpha \subseteq A \rtimes_\alpha G$ is $\cD$-stable.
          \begin{proof}
            This follows from \cite[Theorem 3.3]{HirshbergWinter07}, together with Lemma \ref{lem:fixed-point-in-crossed-product-D-stable}.
          \end{proof}
    \end{prop}


    \begin{defn}
        Let $A$ be a unital, separable C*-algebra. We say that a finite group action $G \act^\alpha A$ has the weak tracial Rokhlin property if for all  $\cF \subseteq A$ finite, $\ee > 0$ and $0 \neq a \in A_+$, there are pairwise orthogonal normalized positive contractions $(e_g)_{g \in G} \subseteq A$ such that
              \begin{enumerate}
                \item $1 - \sum_g e_g \precsim a$;\footnote{For two positive elements $x,y$ in a C*-algebra, we write $x \precsim y$ to mean that $x$ is Cuntz-subequivalent to $y$. That is, there are $(r_n)$ in the C*-algebra such that $r_n^*yr_n \to x$. See \cite[Section 2]{HirshbergOrovitz13}.}
                \item $[e_g,x] \approx_\ee 0$ for all $x \in \cF, g \in G$;
                \item $\alpha_g(e_h) \approx_\ee e_{gh}$ for all $g,h \in G$.
              \end{enumerate}
      \end{defn}

      We note that both Rokhlin and weak tracial Rokhlin actions are necessarily outer. 


      \begin{prop}\label{prop:dsi-wtr-preserves-Z}
          Let $A$ be a unital, simple, separable, nuclear, $\cZ$-stable C*-algebra. If $G \act^\alpha A$ is an action of a finite group with the weak tracial Rokhlin property. Then $A^\alpha \subseteq A \rtimes_\alpha G$ is $\cZ$-stable.
            \begin{proof}
              Let $k \in \bN$. By \cite[Theorem 5.6]{HirshbergOrovitz13} $A \rtimes_\alpha G$ is tracially $\cZ$-absorbing, meaning there are tracially large (in the sense of \cite{TomsWhiteWinter15}) c.p.c. order zero maps $\phi: M_k \to (A \rtimes_\alpha G)_\omega \cap (A \rtimes_\alpha G)'$, which can be chosen to be c.p.c. order zero maps $\phi: M_k \to A_\omega \cap (A \rtimes_\alpha G)'$ by the proof of \cite[Lemma 5.5]{HirshbergOrovitz13}. These tracially large c.p.c. order zero maps yield sequences of positive contractions $c_1 = (c_{1n}),\dots,c_k = (c_{kn}) \in A_\omega \cap (A \rtimes_\alpha G)'$ such that if $(e_n) = e := 1 - \sum_i c_i^*c_i$, we have
\begin{equation}
              \lim_{n \to \omega} \max_{\tau \in T(A)} \tau(e_n) = 0, \inf_m \lim _{n \to \omega} \min_{\tau \in T(A)} \tau(c_{1n}^m) > 0 
\end{equation}
              and $c_ic_j^* = \delta_{ij}c_1^2$. By \cite[Proposition 4.11]{GardellaHirshberg18} (which is much more general, applicable to all countable amenable groups), $A \subseteq A \rtimes_\alpha G$ has equivariant property (SI) since $A$ has property (SI).\footnote{A unital, separable, simple, nuclear, $\cZ$-stable C*-algebra has property (SI) as in \cite{MatuiSato12}} Consequently there exists $s \in A_\omega \cap (A \rtimes_\alpha G)'$ such that $s^*s = 1 - \sum_i c_i^*c_i$ and $c_1s = s$. Altogether,
                \begin{itemize}
                  \item $c_1 \geq 0$;
                  \item $c_ic_j^* = \delta_{ij}c_1^2$;
                  \item $s^*s + \sum_i c_i^*c_i = 1$;
                  \item $c_1s = s$. 
                \end{itemize}
                As mentioned in the proof of $(iv) \Rightarrow (i)$ of \cite{MatuiSato12}, $\cZ_{n,n+1}$ is the universal C*-algebra generated by $n+1$ elements satisfying the above four relations (see \cite[Proposition 5.1]{RordamWinter10} and \cite[Proposition 2.1]{Sato10}), and consequently we have a unital *-homomorphism $\cZ_{n,n+1} \to A_\omega \cap (A \rtimes_\alpha G)'$. Therefore $\cZ \into A_\omega \cap (A \rtimes_\alpha G)'$, giving that the desired inclusion is $\cZ$-stable by Lemma \ref{lem:fixed-point-in-crossed-product-D-stable}.
            \end{proof}
        \end{prop}


        \begin{cor}\label{weaktracialRokhlinintermediate}
            Let $A$ be a unital, simple, separable, nuclear, $\cZ$-stable C*-algebra and $G \act^\alpha A$ be an action of a finite group with the weak tracial Rokhlin property. There exists an isomorphism $\Phi: A \rtimes_\alpha G \simeq (A \rtimes_\alpha G) \otimes \cZ$ such that whenever $C$ is a unital C*-algebra satisfying either
              \begin{enumerate}
                \item $A^\alpha \subseteq C \subseteq A$ or
                \item $A \subseteq C \subseteq A \rtimes_\alpha G$,
              \end{enumerate}
              we have $\Phi(C) = C \otimes \cZ$. 
                \begin{proof}
                  This results from combining Proposition \ref{prop:dsi-wtr-preserves-Z} together with Theorem \ref{Galois}, making note that this is an outer action. 
                \end{proof}
          \end{cor}

        \subsection{The diagonal inclusion associated to a group action}\label{ssection:dse-diagonal-inclusions} In the von Neumann setting, a certain diagonal inclusion associated to several automor\hyp{}phisms was considered in \cite{Popa89,Kawamuro99,Burstein10}, and they play a role in subfactor theory. Here we consider a  unital C*-algebraic inclusion of the same form.

    \begin{defn}
      Let $A$ be a C*-algebra, $\alpha_1,\dots,\alpha_n \in \Aut(A)$. The diagonal inclusion associated to $\alpha_1,\dots,\alpha_n$ is
\begin{equation}
      B(\alpha_1,\dots,\alpha_n) = \left\{\bigoplus_{i=1}^n \alpha_i(a) \mid a \in A \right\} \subseteq M_n(A).
\end{equation}
    \end{defn}

    If $G \act^\alpha A$ is an action of a finite group, we'll write
    \begin{equation}
        B(\alpha) = \left\{\bigoplus_{g \in G} \alpha_g(a) \mid a \in A\right\} \subseteq M_{|G|}(A).
    \end{equation}
We note that a diagonal $B(\alpha) \subseteq M_{|G|}(A)$ is unique up to unitary conjugation (by permutation unitaries). As $\cD$-stability of an inclusion is preserved under unitary conjugation, there is no ambiguity in speaking of $\cD$-stability of the inclusion $B(\alpha) \subseteq M_{|G|}(A)$.
    
    \begin{prop}\label{diagonalinclusion}
        Let $G \act^\alpha A$ be an action of a countable discrete group on a unital, separable C*-algebra. If $G = \langle g_1,\dots,g_n\rangle$, then $A \subseteq A \rtimes_\alpha G$ is $\cD$-stable if and only if
\begin{equation}
        B(\id_A,\alpha_{g_1},\dots,\alpha_{g_n}) \subseteq M_{n+1}(A)
\end{equation}
        is $\cD$-stable.
          \begin{proof}
            First suppose that $A \subseteq A \rtimes_\alpha G$ is $\cD$-stable. Let $\cF \subseteq \cD, \cG \subseteq M_{n+1}(A)$ be finite and $\ee > 0$. Let $\cG' \subseteq A$ be the set of matrix coefficients of elements of $\cG$, together with the identity of $A$, and let $L := \max\{1,\max_{a \in \cG'}\|a\|\}$. Relabel $\id_A,\alpha_{g_1},\dots,\alpha_{g_n}$ as $\alpha_1,\dots,\alpha_{n+1}$. Let
\begin{equation}
            \delta := \frac{\ee}{(4L+1)(n+1)^2}
\end{equation}
            and let $\psi: \cD \to A$ be a u.c.p. $(\cF,\delta,\cG' \cup\{u_{g_i}\}_{i=1}^n)$-embedding, where $(u_g)$ are the implementing unitaries for $\alpha$. Let $\phi: D \to B(\alpha) \subseteq M_{|G|}(A)$ be given by
            \begin{equation}
            \phi(d) := \bigoplus_{i=1}^{n+1} (\alpha_i \circ \psi)(d).
            \end{equation}
            Clearly $\phi$ will be $(\cF,\delta)$-multiplicative since each component is the composition of a *-homomorphism (which are contractive) with a map which is $(\cF,\delta)$-multiplicative. Now for $d \in \cF$ and $a = (a_{ij}) \in \cG$, we have
\begin{equation}
\begin{split}                \|[\phi(d),(a_{ij})]\| &\leq \sum_{i,j=1}^{n+1}\|\alpha_i(\psi(d))a_{ij} - a_{ij}\alpha_j(\psi(d))\| \\
                                       &\leq \sum_{i,j=1}^{n+1}\| \alpha_i(\psi(d))a_{ij} - \psi(d)a_{ij}\| \\
                                       &\ \ \ + \|\psi(d)a_{ij} - a_{ij}\psi(d)\| + \|a_{ij}\psi(d) - a_{ij}\alpha_j(\psi(d))\| \\
                                       &\leq \sum_{i,j=1}^{n+1}\|a_{ij}\|\left(\|\alpha_i(\psi(d)) - \psi(d)\| + \|\psi(d) - \alpha_j(\psi(d))\|\right) \\
                                       &\ \ \  + \|[\psi(d),a_{i,j}]\| \\
                                       &< (n+1)^2(2L(\delta + \delta) + \delta) \\
                                       &= (n+1)^2(4L+1)\delta = \ee.
\end{split}
\end{equation}

              Conversely if the associated diagonal inclusion is $\cD$-stable we note that if $(x_k) \subseteq B(\id_A,\alpha_{g_1},\dots,\alpha_{g_n})$ is central for $M_{n+1}(A)$, writing
              \begin{equation}
              x_k = \bigoplus_{i=1}^{n+1}\alpha_i(a_k)
              \end{equation}
               yields that $(a_k) \subseteq A$ is central for $A$ and is asymptotically fixed by $\alpha_{g_i}, i=1,\dots,n$. In particular if $\cD \into B(\id_A,\alpha_{g_1},\dots,\alpha_{g_n})_\omega \cap (M_{n+1}(A))'$, then $\cD \into A_\omega \cap (A \rtimes_\alpha G)'$. 
          \end{proof}
      \end{prop}

        \begin{cor}
          Let $G \act^\alpha A$ be an action of a finite group on a unital, separable C*-algebra. Then $A \subseteq A \rtimes_\alpha G$ is $\cD$-stable if and only if
\begin{equation}
          B(\alpha) \subseteq M_{|G|}(A)
\end{equation}
          is $\cD$-stable. 
        \end{cor}


  \section{Examples}\label{section:dse-examples}

  \subsection{Non-examples}\label{ssection:dse-non-examples}
  We first start with some non-examples. Villadsen's C*-algebras with perforation will be useful (see \cite{TomsWinter09} for good exposition). Let $\cQ = \bigotimes_n M_n$ denote the universal UHF C*-algebra.

  \begin{theorem}[\cite{Villadsen98,Toms08}]
      There exists a unital, simple, separable, nuclear C*-algebra $C$ satisfying the UCT such that $C \not\simeq C \otimes \cZ$ and $C$ contains the universal UHF algebra unitally. Moreover $C$ is tracial and can be chosen to be AH with
\begin{equation}
       (K_0(A),K_0(A)^+,[1]_0,K_1(A)) = (\bQ,\bQ_+,1,0).
\end{equation}
    \end{theorem}


      \begin{cor}
        There exists an embedding $\cQ \into \cQ$ which is not $\cZ$-stable. In particular, it is not $\cQ$-stable.
          \begin{proof}
            Let $C$ be as above. Note that $\cQ \subseteq C$ so we must find an embedding $C \into \cQ$. As $C$ is unital, separable, exact, satisfies the UCT and has a faithful amenable trace (it has traces, and every such trace will be faithful and amenable since $C$ is nuclear and simple) and there is clearly a morphism between $K_0$-groups, \cite[Theorem D]{Schafhauser20} gives an embedding $C \into \cQ$. Conse\hyp{}quently there is an embedding
\begin{equation}
            \cQ \into C \into \cQ
\end{equation}
            which is not $\cQ$-stable since there is an intermediate C*-algebra $C$ with $C \not\simeq C \otimes \cZ$. 
          \end{proof}
      \end{cor}

        \begin{cor}
          There is an embedding $\cZ \into \cQ$ which is not $\cZ$-stable. 
            \begin{proof}
              Take $C$ as above and take the chain of embeddings (noting that $\cQ$ is $\cZ$-stable)
\begin{equation}
              \cZ \into \cQ \otimes \cZ \simeq \cQ \into C \into \cQ.
\end{equation}
            \end{proof}
        \end{cor}

  \begin{cor}
    There is an embedding $\cZ \into \cO_2$ which is not $\cZ$-stable. 
      \begin{proof}
        Just take the same embedding as above together with an embedding $\cQ \into \cO_2$. 
      \end{proof}
  \end{cor}

    \begin{remark}
      All *-homomorphisms between strongly self-absorbing C*-alge\hyp{}bras are approximately unitarily equivalent by \cite[Corollary 1.12]{TomsWinter07}, or even asymptotically unitarily equivalent by \cite[Theorem 2.2]{DadarlatWinter09}. Therefore $\cD$-stability is not closed under these equivalences (nor homotopy, see \cite[Corollary 3.1]{DadarlatWinter09}).
    \end{remark} 

    The only method we have used to show that an inclusion is not $\cD$-stable is by finding an intermediate algebra which is not $\cD$-stable.
    There are plenty of examples of stably finite C*-algebras with perforation or higher-stable rank (in particular non-$\cZ$-stable C*-algebras \cite{Rordam04}) \cite{Villadsen98, Villadsen99,ElliottVilladsen00,Toms05,HirshbergRordamWinter07,Toms08b,Toms08,TomsWinter09,deLacerdaMortari09,Tikuisis12}. This gives rise to the following two questions.

    \begin{enumerate}
      \item Is there a unital inclusion $B \subseteq A$ of separable C*-algebras such that whenever $C$ is such that $B \subseteq C \subseteq A$, we have $C \simeq C \otimes \cD$ but $B \subseteq A$ is not $\cD$-stable? Is $\cD$-stability equivalent to every intermediate C*-algebra being $\cD$-stable?
      \item To get non-examples we use stably finite C*-algebras with perforation in between sufficiently regular C*-algebras. Is there a way to do this for purely infinite C*-algebras, or is finiteness the only obstruction? Thus we can ask: if $\cD$ is a purely infinite strongly self-absorbing C*-algebra, is every embedding of $\cD$ into itself $\cD$-stable? More specifically, if $B \subseteq A$ is a unital inclusion of simple, separable, purely infinite C*-algebras, is the inclusion $\cO_{\infty}$-stable?
    \end{enumerate}

    Our third question asks if we can get non-examples arising from dynamical systems. 

      \begin{enumerate}
        \item[3.] Is there a unital, separable $\cD$-stable C*-algebra and a (finite) group action $G \act^\alpha A$ such that $A \rtimes_\alpha G$ is $\cD$-stable, but the inclusion is not? One would need $A \rtimes_\alpha G$ to be $\cD$-stable for non-dynamical reasons. 
      \end{enumerate}


  \subsection{Cyclicly permuting tensor powers}\label{ssetcion:dse-cyclic-tensor-powers}

  Here we give a dynamical example to illustrate the discussion in section \ref{section:dse-crossed-products}. In particular, we can look at a consequence of Corollary \ref{autfixedpoint}. 

    \begin{example}
      Let $p,q \in \bN$ be coprime and consider the $q$th tensor power of the UHF algebra $A = M_{p^{\infty}}^{\otimes q}$. Let us examine the action $\bZ_q \act^\sigma A$ given by cyclically permuting the tensors: 
\begin{equation}
      \sigma(a_1 \otimes \cdots \otimes a_q) = a_2 \otimes \cdots \otimes a_q \otimes a_1.
\end{equation}
      One can prove directly or use \cite{HirshbergOrovitz13} or \cite{AGJP22} in order to conclude that this action has the weak tracial Rokhlin property and consequently that $A^{\sigma} \subseteq A \rtimes_\sigma \bZ_q$ is $\cZ$-stable. 

      Alternatively, one can use techniques similar to \cite{HirshbergWinter07} in order to compute the $K$-theory of the fixed point algebra $A^{\sigma}$ to be 
\begin{equation}
      K_0((M_{p^{\infty}}^{\otimes q})^{\sigma}) \simeq \dlim\left(\bZ^q,   \begin{pmatrix}
          p + \frac{p^q - p}{q} & \frac{p^q -p}{q} & \cdots & \frac{p^q - p}{q} \\
          \frac{p^q - p}{q} & p + \frac{p^q - p}{q} & \cdots & \frac{p^q - p}{q} \\
          \vdots & \vdots & \ddots & \vdots \\
          \frac{p^q - p}{q} & \frac{p^q - p}{q} & \cdots & p + \frac{p^q - p}{q}
      \end{pmatrix} \right), 
\end{equation}
      from which one can show that $K_0(A^{\sigma})$ is $p$-divisible.Then using the fact that $K_0(A^{\sigma})$ is $p$-divisible and $A^{\sigma}$ is AF, it follows that  $A^{\sigma}$ is  $M_{p^{\infty}}$-stable. Using Corollary \ref{autfixedpoint}, we then see that $M_{p^{\infty}} \into (A^{\sigma})_\omega \cap A'$. In particular, we have that $A^{\sigma} \subseteq A \rtimes_\sigma \bZ_q$ is $M_{p^{\infty}}$-stable (since clearly if this embedding is fixed by $\bZ_q$, it will commute with the implementing unitaries as well).  
    \end{example}

      \begin{example}
        Following up on the previous example, if we consider the embedding
\begin{equation}
        B := \left\{   \begin{pmatrix}
            x \\ & \sigma(x) \\ & & \ddots \\ & & & \sigma^{q-1}(x)
        \end{pmatrix} \mid x \in M_{p^{\infty}}^{\otimes q} \right\} \subseteq M_q(M_{p^{\infty}}^{\otimes q}) := A,
\end{equation}
        then $B \subseteq A$ is $M_{p^{\infty}}$-stable by Proposition \ref{diagonalinclusion}.
      \end{example}

  \subsection{The canonical inclusion of the CAR algebra in $\cO_2$}\label{ssection:dse-car-in-O2}
    \begin{example}
      Let $\cO_2 = C^*(s_1,s_2)$ be the Cuntz algebra generated by two isometries \cite{Cuntz77}, and consider the inclusion
\begin{equation}
      M_{2^{\infty}} \simeq \ov{\text{span}}\{s_\mu s_\nu^* \mid |\mu| = |\nu|\} \subseteq \cO_2,
\end{equation}
      where for a word $\mu = \{i_1,\dots,i_p\} \in \{1,2\}^p$, $s_\mu = s_{i_1}\cdots s_{i_p}$. This copy of the CAR algebra is precisely the fixed point subalgebra of the gauge action (see \cite{RaeburnBook}). Consider the endomorphism $\lambda: \cO_2 \to \cO_2$ given by
\begin{equation}
      \lambda(x) := s_1xs_1^* + s_2xs_2^*.
\end{equation}
      We note that a sequence $(x_n)$ is $\omega$-asymptotically central for $\cO_2$ if and only if it is $\omega$-asymptotically fixed by $\lambda$. Indeed, if $(x_n)$ is central, then $\|\lambda(x_n) - x_n\| \to^{n \to \omega} 0$ since $[x_n,s_i] \to 0$ for $i=1,2$. On the other hand if $(x_n)$ is asymptotically fixed by $\lambda$, then the inequalities
 \begin{equation}
         \begin{split}
        \|s_ix_n - x_ns_i\| &= \|s_1x_ns_1^*s_i + s_2x_ns_2^*s_i - x_ns_i\| \leq \|\lambda(x_n) - x_n\|\|s_i\| \\
        \|s_i^*x_n - x_ns_i^*\| &= \|s_i^*x_n - s_i^*s_1s_1^* - s_i^*s_2x_ns_2^*\| \leq \|s_i^*\|\|\lambda(x_n) - x_n\|
        \end{split}
 \end{equation}
        imply that $(x_n)$ is asymptotically central. 

        We note that $\lambda|_{M^{2^{\infty}}}$ is the forward tensor shift if we identify $M_{2^{\infty}} = \bigotimes_{\bN} M_2$ (see for example \cite[Section V.4]{DavidsonBook1}). Now \cite{BSKR93} gives an embedding $\xi: M_2 \into (M_{2^{\infty}})$ such that $\lambda_\omega \circ \xi = \xi$. In particular $M_{2^{\infty}} \into (M_{2^{\infty}})_{\omega} \cap \cO_2'$ so that this inclusion is $M_{2^{\infty}}$-stable. 
    \end{example}

    Thinking of $\cO_2$ as the semigroup crossed product $\cO_2 \simeq M_{2^{\infty}} \rtimes_\lambda \bN$ (see \cite{Rordam95,Rordam21}), any intermediate C*-algebra is automatically CAR stable. Consequently each intermediate subalgebra $M_{2^{\infty}} \rtimes d\bN = C^*(M_{2^{\infty}},s_1^d)$ is $M_{2^{\infty}}$-stable. We can do this all concurrently. 

      \begin{cor}
        There exists an isomorphism $\Phi: \cO_2 \simeq \cO_2 \otimes M_{2^{\infty}}$ such that
\begin{equation}
        \Phi(C^*(M_{2^{\infty}},s_1^d)) \simeq C^*(M_{2^{\infty}},s_1^d) \otimes M_{2^{\infty}}
\end{equation}
        for all $d \in \bN$. The same holds if we replace $M_{2^{\infty}}$ by $\cZ$. 
      \end{cor}


  Now let us play with some diagonal inclusions associated to powers of the Bernoulli shift $\lambda$ on $\cO_2$ above.  This will be similar to what was discussed in Section \ref{ssection:dse-diagonal-inclusions}, except we allow endomorphisms.

    \begin{example}
      
  Consider, for $n \in \bN$, the diagonal inclusion
\begin{equation}
  B_n:= \left\{   \begin{pmatrix}
      x \\ & \lambda(x) \\ & & \ddots \\ & & & \lambda^{n-1}(x)
  \end{pmatrix} \mid x \in \cO_2 \right\} \subseteq M_n(\cO_2) =: A_n.
\end{equation}
  Note that both $A_n$ and $B_n$ are isomorphic to $\cO_2$, and in fact this gives an non-trivial inclusion of $\cO_2$ into itself which is $\cO_2$-stable. This is $\cO_2$-stable since a sequence is asymptotically fixed by $\lambda$ if and only if it asymptotically commutes with the algebra. A similar argument to that of Proposition \ref{diagonalinclusion} will yield that this inclusion is $\cO_2$-stable.
    \end{example}

  One can even restrict the diagonal to elements of the CAR algebra $M_{2^{\infty}} \subseteq \cO_2$ sitting as the fixed point subalgebra of the Gauge action as above.
  \begin{example}
  Consider
\begin{equation}
   B_n^{(2)} := \left\{   \begin{pmatrix}
  x \\ & \lambda(x) \\ & & \ddots \\ & & & \lambda^{n-1}(x) \end{pmatrix} \mid x \in M_{2^{\infty}} \right\} \subseteq M_n(\cO_2) = A_n.
\end{equation}
  This is $M_{2^{\infty}}$-stable for the same reasons as above. This gives another inclusion $M_{2^{\infty}} \simeq B_n^{(2)} \subseteq M_n(\cO_2) \simeq \cO_2$ which is CAR-stable.
  \end{example}

  \bibliographystyle{amsalpha}
  \bibliography{biblio}

\end{document} 
